An assessment of carbon emission and sink impacts from biorefining and feedstock choices for producing jet biofuel in repurposed crude refineries

A Natural Resources Defense Council (NRDC) Report

Prepared for the NRDC by Greg Karras, G. Karras Consulting <u>www.energy-re-source.com</u>

August 2021

CONTENTS

	Executive summary, findings and takeaways	page 3
1.	How would refiners rebuild for HEFA jet fuel production?	7
2.	Can refiners make more HEFA jet fuel from some feedstocks than from others?	10
3.	Does switching from one HEFA feedstock to another change processing carbon intensity differently when refiners target jet fuel instead of diesel production?	16
4.	HEFA jet fuel feedstock and carbon sinks: Could the feedstocks that maximize HEFA jet fuel instead of diesel have comparatively high indirect climate impacts?	23
5.	Limitations and suggestions for future work	33
	Data and methods table for feed-specific estimates	37
	Literature cited	44

Executive Summary

Current climate, energy and aviation policy use the term Sustainable Aviation Fuel (SAF) to mean alternatives to petroleum aviation fuel which could include seven types of biofuels and can replace up to half of petroleum jet fuel under existing aviation fuel blending limits. In practice this definition of SAF favors continued use of existing combustion fuel infrastructure to burn a mix of biofuel and petroleum. That is not a net-zero carbon climate solution in itself, and in this sense, SAF is not sustainable. Rather, the partial replacement of petroleum jet fuel with biofuel is meant to incrementally reduce emissions from the hard-to-decarbonize aviation sector and, in concert with more effective measures in other sectors, help to achieve climate stabilization goals.

A question, then, is whether the type of biofuel favored by the existing combustion fuel infrastructure will, in fact, emit less carbon than petroleum. This, the evidence suggests, is a key question for the sustainability of SAF.

Although it is but one proven technology for the production of SAF, Hydrotreated Esters and Fatty Acids (HEFA) technology is the fastest-growing type of biofuel in the U.S. today. This rapid recent and projected growth is being driven by more than renewable fuels incentives. The crucially unique and powerful driver of HEFA biofuel growth is that oil companies can protect troubled and climate-stranded assets by repurposing petroleum crude refinery hydroconversion and hydrogen plants for HEFA jet fuel and diesel biofuels production.

Some HEFA biofuels are reported to emit more carbon per gallon than petroleum fuels. This is in part because HEFA technology depends upon and competes for limited agricultural or fishery yields of certain types—oil crops, livestock fats or fish oils—for its biomass feedstocks. Meeting increased demands for at least some of those feedstocks has degraded natural carbon sinks, causing indirect carbon emissions associated with those biofuels. And it is in part because HEFA feedstocks require substantial hydrogen inputs for HEFA processing, resulting in very substantial direct carbon emissions from fossil fuel hydrogen production repurposed for HEFA biorefining. Both processing strategies, i.e., refining configurations to target jet fuel v. diesel

production, and feedstock choices, e.g., choosing to process palm oil *v*. livestock fat feeds, are known factors in these direct and indirect emissions. That is important because HEFA jet fuel yield is limited, and refiners can use various combinations of feeds and processing strategies to boost jet yield with repurposed crude refining equipment. To date, however, the combined effect of these factors in strategies to boost HEFA jet fuel yield has received insufficient attention.

This report focuses on two questions about climate impacts associated with HEFA jet fuel production in repurposed crude refineries. First, could feedstocks that enable refiners to boost jet fuel yield increase the carbon dioxide emission per barrel—the carbon intensity—of HEFA refining relative to the feeds and processing strategy refiners use to target HEFA diesel yield? Second, could the acquisition of feedstocks that refiners can use to increase HEFA jet fuel yield result in comparatively more serious indirect climate impacts?

The scope of the report is limited to these two questions. Its analysis and findings are based on publicly reported data referenced herein. Data and analysis methods supporting feed-specific original research are given and sourced in an attached data and methods table.¹ Data limitations are discussed in the final chapter. This work builds on recent NRDC-sponsored research² which is summarized in relevant part as context above, and as referenced in following chapters.

Chapter 1 provides an overview of HEFA technology, including the essential processing steps for HEFA jet fuel production and additional options for maximizing jet fuel yield using repurposed crude refining assets. This process analysis shows that a growing fleet of HEFA refineries could, and likely would, use a combination of strategies in which the use of intentional hydrocracking (IHC) could vary widely. HEFA refiners could produce HEFA jet fuel without intentional hydrocracking (No-IHC), produce more HEFA jet fuel with IHC in the isomerization step needed for all HEFA fuels (Isom-IHC), or produce more HEFA jet fuel while shaving the increased hydrogen costs of intentional hydrocracking (Selective-IHC). The strategies chosen would be influenced by the capabilities of crude refineries repurposed for HEFA processing.

Chapter 2 reviews HEFA feedstock limitations and supply options, presents detailed data relating feedstock properties to effects on HEFA jet fuel yields and process hydrogen demand, and ranks individual feedstocks for their ability to increase HEFA jet fuel yield. Differences in chemistry among feeds result in different feed rankings for jet fuel *versus* diesel yields, different feed rankings for increased jet fuel yield among processing strategies, and different feed rankings for hydrogen demand among processing strategies. Palm oil, livestock fats, and fish oils boost jet fuel yield without intentional hydrocracking, and enable more refiners to further boost jet yield with intentional hydrocracking, which increases HEFA process hydrogen demand.

Chapter 3 describes and quantifies refining strategy-specific and feed-specific carbon dioxide (CO₂) emissions from the repurposed crude refinery steam reformers that produce hydrogen for HEFA processing. Feed-specific carbon intensity (CI) rankings for jet fuel-range feed fractions mask those for whole feed actual CI when refiners use the No-IHC process strategy. Refining CI rankings for some feeds with low *v*. high jet yields (e.g., soybean oil *v*.

menhaden fish oil) are reversed in the Selective-IHC strategy compared with the other strategies for increasing HEFA jet fuel yield. Some feeds that increase jet fuel yield have relatively higher process CI (fish oils) while others have relatively lower process CI (palm oil and livestock fats). However, palm oil and livestock fat feeds also enable the highest-CI refining strategies, and all strategies for HEFA jet fuel production result in substantially higher refining CI than the average U.S. petroleum refinery CI. This shows that HEFA jet fuel growth would increase the carbon intensity of hydrocarbon fuels processing.

Chapter 4 reviews natural carbon sinks and assesses potential carbon emission impacts from increasing production of the specific food system resources HEFA refiners can use as feedstocks. Palm oil, livestock, and fisheries production emit from these carbon sinks. Present assessments confirm this "indirect" impact of palm oil biofuels, but suggest livestock fat and fish oil biofuels have relatively low feed production emissions due to the assumption that biofuel demand will not expand livestock production or fisheries catch. Some also assume U.S. policies that discourage palm oil biofuels prevent palm oil expansion to fill in for other uses of biomass biofuels displace. Those assumptions, however, are based on historical data, when biofuels demand was far below total production for the type of biomass HEFA refiners can process. HEFA feedstock demand could far exceed total current U.S. production for all uses of that biomass type—including food and fuel—if HEFA jet fuel replaces as little as 18 percent of current U.S. jet fuel consumption.

With HEFA jet fuel growth to replace 18 percent of U.S. jet fuel, world livestock fat and fish oil production could supply only a fraction of U.S. HEFA feedstock demand unless that demand boosts their production, with consequent indirect carbon impacts. Palm oil production could expand to fill other uses for livestock fat and other plant oils which the increased U.S. biofuel demand would displace. Intensified and expanded production of soybean and other oil crops with relatively high indirect carbon impacts would likely be necessary, in addition, to supply the total demand for both food and fuel. Further, given refiner incentives to repurpose climate-stranded crude refining assets, plausible U.S. HEFA growth scenarios by mid-century range above 18 percent and up to 39 percent of U.S. jet fuel replacement with HEFA jet fuel.

Thus, data and analysis in Chapter 4 suggest the potential for significant indirect carbon emission impacts associated with the mix of HEFA jet fuel feedstocks that could meet plausible future SAF demand, and that high-jet yield feeds could contribute to or worsen these impacts.

Crucially, causal factors for these impacts would be inherent and mutually reinforcing. HEFA technology repurposed from crude refineries can process only feedstocks that are coproduced from food resources, it requires large hydrogen inputs that boost refining emissions to marginally improve its low jet fuel yield, and even then, it could require more than two tons of carbon-emitting feedstock production per ton of HEFA jet fuel produced.

Findings and takeaways from this work follow below.

Findings and Takeaways

Finding 1.	Hydrotreated Esters and Fatty Acids (HEFA) biofuel technology has inherent limitations that affect its potential as a sustainable aviation fuel: low jet fuel yield on feedstock, high hydrogen demand, and limited sustainable feedstock supply.
Takeaway	Climate-safe plans and policies will need to prioritize alternatives to petroleum jet fuel combustion which do not have known sustainability limitations.
Finding 2.	Switching HEFA feedstocks to target increased jet fuel yield could increase the carbon intensity—CO ₂ emitted per barrel feed—of HEFA refining, compared with targeting HEFA diesel yield. HEFA refining carbon intensity could increase in 80 percent of plausible feed switch and processing combinations targeting jet fuel. Direct emission impacts could be significant given that the carbon intensity of HEFA refining substantially exceeds that of U.S. petroleum refining.
Takeaway	Environmental impact assessments of proposed HEFA projects will need to address potential emissions from future use of HEFA refineries to maximize jet fuel production, and assess lower emitting alternatives to repurposing existing high-carbon refinery hydrogen plants.
Finding 3.	One of three feeds that could boost HEFA jet fuel yield causes carbon emissions from deforestation for palm plantations, and the other two cannot meet potential HEFA feedstock demand without risking new carbon emissions from expanded livestock production or fisheries depletion. These indirect impacts could be significant given that feedstock demand for replacing only a small fraction of current U.S. jet fuel with HEFA jet fuel would exceed total U.S. production of HEFA feedstocks biomass—biomass which now is used primarily for food.
Takeaway	Before properly considering approvals of proposed HEFA projects, permitting authorities will need to assess potential limits on the use of feedstocks which could result in significant climate impacts.
Finding 4.	Natural limits on total supply for the type of feedstock that HEFA technology can process appear to make replacing any significant portion of current petroleum jet fuel with this type of biofuel unsustainable.
Takeaway	Sustainable aviation plans will need to consider proactive and preventive limits on HEFA jet fuel, in concert with actions to accelerate development and deployment of sustainable, climate-safe alternatives.

1. How would refiners rebuild for HEFA jet fuel production?

Oil companies can repurpose existing fossil fuel hydrogen plants, hydrocrackers, and hydrotreaters at their petroleum refineries to produce jet fuel and diesel biofuels using a technology called hydrotreating esters and fatty acids (HEFA). "Hydrotreating" means a hydro-conversion process: the HEFA process reacts biomass with hydrogen over a catalyst at high temperatures and pressures to form hydrocarbons and water. "Esters and fatty acids" are the type of biomass this hydro-conversion can process: the triacylglycerols and fatty acids in plant oils, animal fats, fish oils, used cooking oils, or combinations of these biomass lipids.¹

HEFA processing requires a sequence of steps, performed in separate hydro-conversion reactors, to deoxygenate and isomerize (restructure) the lipids feedstock, and very substantial hydrogen inputs for those process steps, in order to produce diesel and jet fuels.²

One problem with using HEFA technology for Sustainable Aviation Fuel (SAF) is that these hydrodeoxygenation and isomerization steps alone can convert only a fraction of its feedstock into jet fuel—as little as 0.128 pounds of jet fuel per pound of soybean oil feed.³ Intentional hydrocracking can boost HEFA jet fuel yield to approximately 0.494 pounds per pound of feed,³ however, that requires even more hydrogen, and can require costly additional refining capacity. This chapter describes the range of processing strategies that refiners could use to increase HEFA jet fuel yields from their repurposed crude refineries.

1.1 Step 1: Hydrodeoxygenation (HDO) of jet fuel (and diesel) hydrocarbons

HEFA processing produces diesel and jet fuels from the hydrocarbon chains of fatty acids. In all HEFA feedstocks, fatty acids are bound in triacylglycerols that contain substantial oxygen, and various numbers of carbon double bonds. To free the fatty acids and make fuels that can burn like petroleum diesel and jet fuel from them, that oxygen must be removed from the whole feed. This first essential step in HEFA processing is called hydrodeoxygenation (HDO).

HDO reaction chemistry is complex, as reviewed in more detail elsewhere,² and its intended reactions all consume hydrogen by forcing it into the feedstock molecules. Process reactions insert hydrogen to free fatty acids from triacylglycerols ("depropanation") and to remove oxygen by bonding it with hydrogen to form water ("deoxygenation"). And along with those reactions, still more hydrogen bonds with the carbon chains to "saturate" the carbon double bonds in them. These reactions proceed at high temperatures and pressures in the presence of a catalyst to yield the intended HDO products: deoxygenated hydrocarbon chains which can be further processed to make diesel and jet fuels.

1.2 Step 2: Isomerization of jet fuel and diesel hydrocarbons

Isomerization restructures the saturated straight-chain hydrocarbons produced by HDO, which are too waxy to burn well or safely in diesel or jet engines, by turning these straight-chain hydrocarbons into their branched-chain isomers. This is the second essential HEFA process step.

Like HDO, isomerization reactions are complex, proceed at high temperatures and pressures in the presence of a catalyst, and require substantial hydrogen inputs.² However, isomerization process reactions, conditions, and catalysts differ substantially from those of HDO and, instead of consuming the hydrogen input as in HDO, most of the hydrogen needed for isomerization can be recaptured and recycled.² These differences have so far required a separate isomerization processing step, performed in a separate process reactor, to make HEFA diesel and jet fuel.

1.3 Additional option of intentional hydrocracking (IHC)

Hydrocracking breaks ("cracks") carbon bonds by forcing hydrogen between bonded carbon atoms at high temperature and pressure. This cracks larger hydrocarbons into smaller ones. It is an unwanted side reaction in HDO and some isomerization processing since when uncontrolled, it can produce compounds too small to sell as either diesel or jet fuel. *Intentional* hydrocracking (IHC) uses specialized catalysts and process conditions different from those required by HDO to crack HDO outputs into hydrocarbons in the jet fuel range.

Thus, while HEFA refiners can make jet fuel with HDO and isomerization alone (No-IHC), they could make more jet fuel by adding IHC to their processing strategy. Adding IHC for the HDO output can boost jet fuel yield to approximately 49.4 percent of HEFA feedstock mass (49.4 wt.%).³ This boost is important, compared with No-IHC jet fuel yield of approximately 12.8 wt.% on soybean oil,³ the most abundant HEFA feedstock produced in the U.S.² However, hydrocrackers are expensive to build for refineries that do not already have them,⁴ and IHC increases demand for hydrogen plant production capacity by approximately 1.3 wt.% on feed (800 cubic feet of H₂/barrel).^{2 3} New capacity for additional hydrogen production is also costly to refiners that cannot repurpose existing capacity. HEFA refiners that choose the IHC option to maximize jet fuel yield might choose one processing strategy to minimize new hydrocracking capacity cost, or another processing strategy to minimize new hydrogen capacity cost.

1.3.1 IHC in isomerization process units

Hydrocracking and isomerization can be accomplished in a repurposed crude refinery hydrocracker, given the necessary retooling and catalyst for HEFA HDO output processing.² Thus, a crude refinery with sufficient existing hydrocracking and hydrogen capacity for the whole HEFA feed stream it plans to process could repurpose that equipment for IHC in the isomerization step of its repurposed HEFA process configuration. This "Isom-IHC" processing strategy would allow that refiner to maximize HEFA jet fuel yield without the capital expense of building a new hydrocracker. However, combining intentional hydrocracking in isomerization, which is required for all HEFA fuels, cracks the entire output from the HDO step, incurring the 800 cubic feet of hydrogen per barrel cost increment on the entire HEFA feed. If a refiner lacks the existing hydrogen capacity, Isom-IHC could entail building new hydrogen plant capacity.

1.3.2 Selective IHC in separate hydrocracking process units

HEFA refiners separate the components of their HDO and isomerization outputs to re-run portions of the feed through those processes and to sell HEFA diesel and jet fuel as separate products. That distillation, or "fractionation," capacity could be used to separate the jet fuel produced by HDO and isomerization processing from their hydrocarbons output, and feed only those hydrocarbons outside the jet fuel range to a separate intentional hydrocracking unit. This "Selective-IHC" processing strategy could increase jet fuel yield while reducing IHC hydrogen consumption, and new hydrogen plant costs, compared with those of the Isom-IHC strategy. However, it would not eliminate the hydrogen production cost of IHC, and more importantly for refiners that lack the existing hydrocracking capacity before repurposing their crude refineries, it would entail building expensive new hydrocrackers.

1.4 Three potential HEFA jet fuel processing strategies

HEFA feedstock supply limitations,² differences in hydrogen production and hydrocracking capacities among U.S. refineries,⁵ and the differences between processing strategies described above suggest the broad outlines of a prospective future HEFA jet fuel refining fleet. Refiners that can repurpose sufficient capacity could maximize HEFA jet fuel yield using IHC strategies. The fleet-wide mix would be influenced initially by whether existing hydrocracking or hydrogen production capacity would limit total production by each refinery to be repurposed. Later, the relative costs of hydrogen production *v*. hydrocracking could affect the mix of Selective-IHC *v*. Isom-IHC in the mid-century HEFA refining fleet.

Refiners that lack sufficient capacity for IHC could repurpose for the No-IHC strategy and coproduce HEFA jet fuel along with larger volumes of HEFA diesel. Then, increasing costs of the much higher feed volume needed per gallon of HEFA jet fuel yield from the No-IHC strategy could limit this strategy to a small portion of the refining fleet by mid-century. Declining HEFA diesel demand, as electric and fuel cell vehicles replace diesel vehicles, could further drive this this limitation of the No-IHC processing strategy. However, refiners that do not use intentional hydrocracking could seek to boost HEFA jet fuel yield in another way.

2. Can refiners make more HEFA jet fuel from some feedstocks than from others?

HEFA biofuel technology is limited to a particular subset of world biomass supply for its feedstock. Despite that limitation, however, differences among these lipid feeds could affect both HEFA processing and jet fuel yield. This chapter assesses individual HEFA feedstocks for potential differences in HEFA processing and HEFA jet fuel yield.

Results reveal strong interactions between feedstock and processing configuration choices. In essential HEFA process steps, feed choices affect jet fuel yield and hydrogen demand, both of which affect options to further boost jet yield with intentional hydrocracking. Both feedstock and processing choices can increase hydrogen demand, which can affect processing to boost jet fuel yield where hydrogen supply is limited. Feed-driven and process strategy-driven impacts on hydrogen demand overlap, however, feed rankings for hydrogen differ from those for jet yield, and differ among processing configurations. From the lowest to highest impact combinations of feedstock and processing options, jet fuel yield and hydrogen demand increase dramatically.

Palm oil, livestock fat, and fish oil have relatively high jet fuel yields without intentional hydrocracking, and relatively high potentials to enable further boosting jet fuel yields with intentional hydrocracking (IHC).

2.1 HEFA feedstock limitations and supply options

HEFA biofuel technology relies on the fatty acids of triacylglycerols in biomass lipids for its feedstocks, as described in Chapter 1. Sources of these in relevant concentrations and quantities are limited to farmed or fished food system lipids resources. Among its other problems, which are addressed in a subsequent chapter, this technological inflexibility limits feedstock choices for refiners seeking to increase HEFA jet fuel yield.

Historically used lipid biofuel feedstock supplies include palm oil, soybean oil, distillers corn oil, canola (rapeseed) oil, and cottonseed oil among the significant HEFA oil crop feeds; livestock fats, including beef tallow, pork lard, and poultry fats; and fish oils—for which we

analyze data on anchovy, herring, menhaden, salmon, and tuna oils.¹ Additionally, though it is a secondary product from various mixtures of these primary lipid sources, and its supply is too limited to meet more than a small fraction of current HEFA demand,² we include used cooking oil (UCO) in our analysis.¹

2.2 Feedstock properties that affect HEFA jet fuel production

2.2.1 Feedstock carbon chain length

Jet fuel is a mixture of hydrocarbons that are predominantly in the range of eight to sixteen carbon atoms per molecule. In fuel chemistry shorthand, a hydrocarbon with 8 carbons is "C8" and one with 16 carbons is "C16," so the jet fuel range is C8–C16. Similarly, a fatty acid chain with 16 carbons is a C16 fatty acid. Thus, since fuels produced by the essential HEFA process steps—hydrodeoxygenation (HDO) and isomerization—reflect the chain lengths of fatty acids in the feed,² the ideal HEFA jet fuel feed would be comprised of C8–C16 fatty acids. But there is no such HEFA feedstock.

In fact, the majority of fatty acids in HEFA lipids feeds, some 53% to 95% depending on the feed, have chain lengths outside the jet fuel range.¹ This explains the low jet fuel yield problem with relying on HEFA technology for Sustainable Aviation Fuel (SAF) described in Chapter 1. However, that 53–95% variability among feeds also reveals that refiners could make more HEFA jet fuel from some HEFA feedstocks than from others.

2.2.2 Feedstock-driven process hydrogen demand

Options to increase HEFA jet fuel yield using intentional hydrocracking could be limited by hydrogen supplies available to refiners, and HDO, an essential HEFA process step, consumes hydrogen to saturate carbon double bonds in feeds and remove hydrogen from them (Chapter 1). HDO accounts for the majority of HEFA process hydrogen demand, and some HEFA feeds have more carbon double bonds, somewhat higher oxygen content, or both, compared with other HEFA feeds.² Thus, some HEFA feeds consume more process hydrogen, and thereby have more potential to affect jet fuel yield by limiting high-yield processing options, than other feeds.

2.3 Ranking HEFA feedstocks for jet fuel production

2.3.1 Effects on HDO yield

Table 1 summarizes results of our research for the chain length composition of fatty acids in HEFA feedstocks.¹ This table ranks feeds by their jet fuel range (C8–C16) fractions. Since fuels produced by the essential HDO and isomerization steps in HEFA processing reflect the chain lengths of HEFA feeds, the volume percentages shown in Table 1 represent potential jet fuel yield estimates for the processing strategy without intentional hydrocracking (No-IHC).

	Jet fuel fraction (C8–C16) (volume % on whole feed)	Diesel fraction (C15–C18) (vol. %)	> C16 (vol. %)	>C18 (vol. %)
Palm oil	46.5	95.6	53.5	0.5
Menhaden oil	42.3	59.8	57.7	31.2
Tallow fat	33.3	95.2	66.7	0.4
Herring oil	32.7	49.3	67.3	42.7
Poultry fat	32.7	98.1	67.3	1.1
Anchovy oil	32.6	52.2	67.4	40.9
Tuna oil	31.5	48.9	68.5	44.5
Lard fat	30.0	96.5	70.0	2.1
Salmon oil	27.5	49.7	72.5	44.0
UCO 10 th P.*	26.8	97.9	73.2	1.1
Cottonseed oil	25.7	98.7	74.3	0.4
Corn oil (DCO)*	13.6	98.9	86.4	1.1
UCO 90th P.*	12.9	99.2	87.1	0.8
Soybean oil	11.7	99.5	88.3	0.4
Canola oil	4.8	96.8	95.2	3.1
Yield-wtd. Average	26.3	97.4	73.7	1.0

Table 1. Chain length' composition of fatty acid chains in HEFA feedstocks, ranked by jet fuel fraction.

*Cx: fatty acid chain of x carbons. UCO: used cooking oil. 10th P.: 10th Percentile. DCO: Distillers corn oil. Data from Table 8, except world yield data by feed type for yield-weighted average shown from Table 7. Percentages do not add; fractions overlap.

Potential feed-driven effects on jet fuel yield shown in Table 1 range tenfold among feeds, from approximately 4.8% on feed volume for canola oil to approximately 46.5% for palm oil. For context, since supplies of some feeds shown are relatively low, it may be useful to compare high jet fuel yield feeds with soybean oil, the most abundant HEFA feed produced in the U.S.² Palm oil, the top ranked feed for jet fuel yield, could potentially yield nearly four times as much HEFA jet fuel as soybean oil, while menhaden fish oil and tallow might yield 3.6 times and 2.8 times as much jet fuel as soy oil, respectively. Again, this is for the No-IHC processing strategy.

2.3.2 Effects on IHC strategies yields

Feed-driven jet fuel yield effects could allow intentional hydrocracking (IHC) to further boost HEFA jet fuel yield, depending on the IHC processing strategy that refiners may choose. At 49.4 wt.% on feed (Chapter 1), or approximately 58 volume percent given the greater density of the feed than the fuel, IHC jet fuel yield exceeds those of the feed-driven effects shown in Table 1. But IHC adds substantially to the already-high hydrogen demand for essential HEFA process steps (Chapter 1). In this context, the eight highest-ranked feeds for jet fuel yield in Table 1 may allow a refiner without the extra hydrogen supply capacity to use IHC on its entire feed to use Selective-IHC on 53.5% to 70% of its feed. This indirect effect of feed-driven jet fuel yield on process configuration choices has the potential to further boost HEFA jet fuel yield.

Direct feedstock-driven effects on process hydrogen demand, which can vary by feed as described above, must be addressed along with this indirect effect. <u>See</u> Table 2 below.

Feedstock	Jet fraction (C8–C16) ^a		Diesel fraction	Diesel fraction (C15–C18) ^a		Longer chains (> C18) ^{a b}	
grouping	HDO kg/b ^c	Sat kg/bd	HDO kg/b ^c	Sat kg/bd	HDO kg/b ^c	Sat kg/bd	
High jet/high diesel							
Palm oil	4.38	< 0.01	4.77	0.64	3.52	0.15	
Tallow fat	4.53	0.14	4.70	0.62	3.62	0.19	
Poultry fat	4.58	0.25	5.04	0.92	3.99	0.67	
Lard fat	4.43	0.11	4.84	0.75	5.39	1.68	
UCO (10 th Pc.)	4.52	0.20	5.02	0.92	4.30	0.75	
Cottonseed oil	4.30	0.02	5.47	1.34	3.51	0.16	
High jet/low diesel							
Menhaden oil	4.72	0.28	5.07	0.85	8.64	4.83	
Herring oil	4.77	0.30	5.09	0.89	6.11	2.52	
Anchovy oil	4.72	0.28	5.22	1.02	8.07	4.31	
Tuna oil	4.67	0.24	4.81	0.64	8.06	4.34	
Salmon oil	4.51	0.09	5.18	1.01	7.99	4.27	
Low jet/high diesel							
Corn (DCO) oil	4.27	0.01	5.60	1.48	4.87	1.38	
UCO (90 th Pc.)	4.35	0.09	5.56	1.45	3.38	0.00	
Soybean oil	4.28	0.01	5.70	1.59	3.31	0.00	
Canola oil	4.35	0.07	5.45	1.37	3.98	0.55	

Table 2. Hydrogen demand for hydrodeoxygenation (HDO) of HEFA feedstocks, grouped by HDO jet fuel and diesel hydrocarbon yields. Data in kilograms hydrogen per barrel of feed fraction (kg H₂/b)

a. Feedstock component fractions based on carbon chain lengths of fatty acids in feeds. **b.** Fatty acid chains with more than 18 carbons (> C18), which might be broken into two hydrocarbon chains in the jet fuel range (C8–C16) by intentional hydrocracking (IHC). **c.** HDO: hydrodeoxygenation; hydrogen consumed in HDO reactions, including saturation. **d.** Sat: saturation, H₂ needed to saturate carbon double bonds in the feedstock component, included in HDO total as well and broken out here for comparisons between types of feeds. <u>See</u> Table 8 for details of data, methods, and data sources. Note that fatty acids with 15–16 carbons (C15–C16) are included in both the jet fuel and the diesel fuel ranges. **UCO:** Used cooking oil, a highly variable feed; the 10th and 90th percentiles of this range of variability are shown.

2.3.3 Effects on process hydrogen demand

Table 2 shows process hydrogen demand for HDO, and the portion of HDO accounted for by saturation of carbon double bonds, for fractions of each feedstock. The important detail this illustrates is that saturation of carbon double bonds—especially in the larger-volume diesel fraction and, for fish oils, the longer chain fraction—explains most of the differences in direct effects on hydrogen demand among feeds. At less than 1% to more than half of HDO hydrogen demand, saturation drives differences in hydrogen demand among feed fractions (Table 2). Further, these differences peak in the diesel and longer chain fractions of feeds (*Id.*), and the combined volumes of these diesel and longer chain fractions are both high for all feeds and variable among feeds (Table 1).

Since HDO is an essential step in all HEFA processing strategies (Chapter 1), this evidence that process hydrogen demand varies among feeds because of the processing characteristics of whole feeds means we can compare hydrogen demand across processing strategies based on whole feeds. Table 3 shows results from this comparison across processing strategies.

Feedstock grouping	No-IHC ª (kg H₂/b)	Selective-IHC ^b (kg H ₂ /b)	lsom-IHC ⁰ (kg H₂/b)
High jet/high diesel			
Palm oil	4.79	5.79	6.60
Tallow fat	4.71	6.11	6.70
Poultry fat	5.03	6.28	6.85
Lard fat	4.85	6.13	6.65
UCO (10 th P.)	5.01	6.37	6.83
Cottonseed oil	5.44	6.84	7.28
High jet/low diesel			
Menhaden oil	6.18	7.30	8.02
Herring oil	5.50	6.76	7.33
Anchovy oil	6.37	7.67	8.23
Tuna oil	6.29	7.62	8.16
Salmon oil	6.40	7.78	8.25
Low jet/high diesel			
Corn (DCO) oil	5.58	7.19	7.42
UCO (90 th P.)	5.55	7.17	7.39
Soybean oil	5.68	7.33	7.52
Canola oil	5.40	7.16	7.24
Feed-wtd. Average	5.24	6.62	7.07

Table 3. Hydrogen demand in the no intentional hydrocrack	king (No-IHC), Selective IHC and Isom-IHC
processing strategies by feed grouping and feed.	ka H ₂ /b: kiloarams hvdrogen/barrel whole feed

a. Intentional hydrocracking (IHC) is not used. **b.** Intentional hydrocracking (IHC) is selective because in this strategy HDO output is separately isomerized, and only the non-jet fuel hydrocarbons from HDO are fed to IHC. **c.** Isomerization and IHC are accomplished in the same process step in this strategy; all HDO output, including the jet fuel fraction, is fed to intentional hydrocracking in this strategy. <u>See</u> Table 8 for details of data, methods, and data sources;¹ Table 7 for world feed data used to derive feed-weighted averages. **UCO:** Used cooking oil, a highly variable feed; 10th and 90th percentiles of range shown.

2.3.4 Interactions between feedstock and processing choices

Feedstock and process strategy choices combined can impact HEFA process hydrogen demand dramatically (Table 3). As expected, IHC increases hydrogen demand for all feeds, however, feed-driven and process strategy-driven effects overlap. The maximum feed-driven impact in the No-IHC strategy (6.40 kg H₂/b) exceeds the minimum (5.79 kg H₂/b) in the Selective-IHC strategy (*Id.*). Similarly, the maximum feed-driven impact in the Selective-IHC strategy (*Id.*). Similarly, the maximum feed-driven impact in the Selective-IHC strategy (*Id.*). Hydrogen demand increases by approximately 75% from the lowest impact (4.71 kg H₂/b) to the highest impact (8.25 kg H₂/b) combination of feedstock and processing strategy (*Id.*).

Feed rankings for hydrogen demand differ from feed rankings for jet fuel yield (tables 1, 3). Palm oil ranks at the top for jet fuel yield and at or near the bottom for hydrogen demand while in contrast, fish oils are among the highest ranked feeds for both jet yield and hydrogen demand. Livestock fats are among the highest ranked feeds for jet fuel yield and among the lowest ranked feeds for hydrogen demand. The lowest ranked feeds for jet fuel yield, soybean and canola oils, are medium-ranked to high-ranked feeds for hydrogen demand.

Relatively lower hydrogen demand for palm oil and livestock fats across the columns in Table 3 further illustrates how interactions of feedstock and processing strategies can contribute to increased jet fuel yields. For example, the relative Isom-IHC hydrogen demand reduction achievable by switching from soybean oil to tallow (-0.82 kg/b; -10.9%) or from soybean oil to palm oil (-0.92 kg/b; -12.2%) can help to support the highest jet fuel yield processing strategy in situations where refinery hydrogen production capacity is marginally limited.

Results in Table 3 also reveal that some feedstocks switch rankings between the Selective-IHC strategy and other processing strategies. In one example, canola oil feedstock demands more hydrogen than cottonseed oil feedstock for Selective-IHC but slightly less than cottonseed oil for the No-IHC and Isom-IHC strategies (Table 3). This corresponds to the greater fraction of canola oil than cottonseed oil sent to intentional hydrocracking for the Selective-IHC strategy (*see* Table 1, > C16 vol. %).

Another example: Only some 57.7% of the total Menhaden oil feed volume goes to intentional hydrocracking for Selective-IHC, as compared with 88.3% of the soybean oil feed (*Id.*). Consequently, Menhaden oil demands less hydrogen than soybean oil for Selective-IHC but more hydrogen than soybean oil for the other processing strategies (Table 3).

Putting these direct and indirect feed-driven effects together, consider switching from soybean oil to tallow for Selective-IHC at a 50,000 to 80,000 b/d refinery—which is in the range of projects now proposed in California.² The direct effect on HDO from this soy oil-to-tallow switch, shown in the No-IHC column of Table 3 (-0.97 kg H₂/b), carries over to Selective-IHC. The indirect effect sends 21.6% less of the total tallow feed to hydrogen-intensive cracking for Selective IHC than that of soy oil (Table 1, > C16 fractions), further boosting hydrogen savings from the switch to -1.22 kg/b on total feed (Table 3). At feed rates of 50,000–80,000 b/d, this might save the refiner construction and operating costs for 61,000 to 97,600 kg/d of hydrogen capacity. Expressed as volume in millions of standard cubic feet per day (MMSCFD), that is the equivalent of a 24 to 38 MMSCFD hydrogen plant.

At the same time that switching from soy with No-IHC to tallow with Selective-IHC could enable the higher-yield processing strategy, however, net process hydrogen demand would increase by 0.43 kg/b (Table 3), an increase in this example of 8.4 to 13.5 MMSCFD.

Thus, examining feed and processing interactions reveals that switching to feeds with higher jet-range fractions, lower HDO hydrogen demand, or both enables refiners with limited hydrogen supplies to use intentional hydrocracking and thereby further boost jet fuel yields. More broadly, these results show refiners can make more HEFA jet fuel from some feedstocks than from others, but that doing so could result in substantially increased hydrogen demand for some combinations of feedstock and processing choices.

3. Does switching from one HEFA feedstock to another change processing carbon intensity differently when refiners target jet fuel instead of diesel production?

Switching feedstocks and production targets can affect the per-barrel emissions—the *carbon intensity*—of HEFA refining dramatically. The vast majority of direct CO₂ emission from HEFA refining emits from petroleum refinery steam reformers that refiners repurpose to supply HEFA process hydrogen demand.² The reformer emissions further increase with increasing hydrogen production.² As shown in Chapter 2, refiners could switch feeds to boost HEFA jet fuel yield in ways that increase refinery hydrogen demand differently compared with targeting HEFA diesel yield. This chapter evaluates the carbon intensity (CI) impacts of HEFA refining that could result from targeting HEFA jet fuel yield instead of diesel yield, and weighs their significance against the CI of petroleum refining.

3.1 CO2 co-production and emission from hydrogen production by steam reforming

3.1.1 How steam reforming makes hydrogen

Steam reforming is a fossil fuel hydrogen production technology that co-produces CO₂. The process reacts a mixture of superheated steam and hydrocarbons over a catalyst to form hydrogen and CO₂. Hydrocarbons used include methane from natural gas, and it is often called steam methane reforming (SMR), but crude refiners use hydrocarbon byproducts from refining such as propane, along with methane from purchased natural gas, as feeds for the steam reformers that they could repurpose for HEFA processing.

3.1.2 How steam reforming emits CO₂

Both its CO_2 co-product and CO_2 formed in its fuel combustion emit from steam reforming. An energy-intensive process, steam reforming burns fuel to superheat process steam and feed, and burns more fuel for energy to drive pumps and support process reactions. Steam reforming fuel combustion emissions are reformer-specific and vary by plant. Based on verified permit data for 11 San Francisco Bay Area crude refinery steam reforming plants, we estimate median

fuel combustion emissions of approximately 3.93 grams of CO_2 emitted per gram of hydrogen produced (g CO_2/g H₂), conservatively assuming methane fuel.² Co-product emissions are larger still, and vary by feed, with approximately 5.46 g CO_2/g H₂ emitting from methane feed and 6.56 g CO_2/g H₂ emitting from propane feed.² The coproduct and combustion emissions are additive.

3.1.3 Steam reforming CO₂ emission estimate

HEFA refinery steam reforming can be expected to use a feed and fuel mix that includes the propane byproduct from the process reactions discussed in Chapter 1 and natural gas methane. Based on process chemistry we conservatively assume 79% methane/21% propane feed with 100% methane fuel. From these figures we estimate typical HEFA steam reforming emissions of approximately 9.82 g CO₂/g H₂. This estimate is for repurposed crude refinery steam reformers, which are aging and may not be as efficient as newer steam reformers.² For context, however, our estimate is within 2.5% of a recent independent estimate of median emissions from newer merchant steam methane reforming plants, when compared on a same-feed basis.²

Thus, repurposed refinery steam reforming emits CO_2 at nearly ten times its weight in hydrogen supplied. With the high hydrogen demand for HEFA processing shown in Chapter 2, that is a problem. Since steam reforming emissions increase with increased production to meet increased hydrogen demand, the refining CI values reported below are based on the emission factor described above (9.82 g CO_2/g H₂) and the hydrogen demand data from Chapter 2.

3.2 Feedstock effects on CI resulting from HDO hydrogen demand

Hydrodeoxygenation (HDO) is an essential step, and is the major hydrogen consuming step, in all HEFA processing strategies (chapters 1 and 2). The data in Table 4 represent the HEFA processing strategy that uses HDO without intentional hydrocracking (No-IHC).

3.2.1 Feedstock HDO chemistry impact on HEFA refining CI

Table 4 shows effects of feedstock HDO chemistry on HEFA steam reforming emissions. Steam reforming-driven CI (kg/b: kg CO₂ per barrel feed) is substantially higher for whole feeds than for their jet fuel fractions. This is because the non-jet fractions need more hydrogen to saturate carbon double bonds and their combined volumes are larger than that of the jet fuel fraction (tables 1 and 2). Further, the extent of these differences between fractions varies among feeds (*Id.*). This is why feeds change ranks between the columns in Table 4. For example, the jet fuel fraction of palm oil has higher CI than that of soybean oil even though the whole feed data show that soybean oil is a higher CI feed. This variability among feed fractions also is why fish oil CI is high for both the jet fraction and the whole feed.

3.2.2 Need to account for whole feed impact

Does Table 4 show that palm oil could be a higher refining CI feed than soybean oil? No. Since the HDO step is essential for removing oxygen from the whole feed to co-produce both HEFA jet fuel and HEFA diesel, choosing any feed results in the CI impact of that whole feed.

Jet fuel fraction	n (C8–C16)	Whole feed (≥ C8)			
Feed (rank)	CO ₂ (kg/b feed)	Feed (rank)	CO ₂ (kg/b feed)		
Herring oil	46.8	Salmon oil	62.8		
Menhaden oil	46.4	Anchovy oil	62.5		
Anchovy oil	46.4	Tuna oil	61.7		
Tuna oil	45.9	Menhaden oil	60.7		
Poultry fat	45.0	Soybean oil	55.8		
Tallow fat	44.5	Distillers corn oil	54.8		
UCO (10 th Percentile)	44.4	UCO (90 th Percentile)	54.4		
Salmon oil	44.3	Herring oil	54.0		
Lard fat	43.5	Cottonseed oil	53.4		
Palm oil	43.0	Canola oil	53.1		
Canola oil	42.7	Poultry fat	49.4		
UCO (90 th Percentile)	42.7	UCO (10 th Percentile)	49.2		
Cottonseed oil	42.2	Lard fat	47.6		
Soybean oil	42.0	Palm oil	47.1		
Distillers corn oil	41.9	Tallow fat	46.2		

Table 4. Hydrogen steam reforming emissions associated with the jet fuel fraction *v*. whole HEFA feeds in the HDO (No IHC) refining strategy; comparison of feed ranks by emission rate.

C8–C16: fatty acid chains with 8 to 16 carbon atoms. \ge **C8:** fatty acid chains with 8 or more carbon atoms. **Menhaden:** a fish. **UCO:** used cooking oil, a variable feed; 10th and 90th percentiles shown. Data from Table 2 at 9.82 g CO₂/g H2 steam reforming.

While the jet fuel fraction data in this table helps to inform why feed quality impacts refining CI, we need to account for those CI impacts of whole feeds shown in Table 4.

3.2.3 High-jet feeds can increase or decrease HDO-driven CI

HDO-driven CI findings for whole feeds reveal mixed CI results for high-jet fuel yield feedstocks in No-IHC processing. Fish oils rank highest for steam reforming-driven CI while livestock fats and palm oil rank lowest (Table 4). Thus, for this processing strategy, switching feeds to boost jet fuel yield can increase or decrease refining CI. However, No-IHC also is the processing strategy that HEFA refiners use to maximize diesel yield rather than jet fuel yield. Feedstock quality interacts with other processing choices in different ways that could further boost HEFA refining CI along with jet fuel yield, as shown below.

3.3 Feedstock effects on CI resulting from Selective-IHC hydrogen demand

3.3.1 Process strategy impact of high-jet feeds

High jet yield feeds result in less input to Selective-IHC, enabling marginally hydrogenlimited refiners to further boost jet fuel yield via Selective-IHC, but this requires additional hydrogen (chapters 1 and 2). Intentional hydrocracking (IHC) thus increases hydrogen steam reforming rates and emissions, increasing refining CI for all feeds, as shown in Table 5. This impact overlies the HDO impact, so that feed CI values overlap between columns. For example, the tuna oil No-IHC CI (61.7 kg/b) exceeds the tallow Selective-IHC CI (60.0 kg/b), and the anchovy oil Selective-IHC CI (75.3 kg/b) exceeds the soy oil Isom-IHC CI (73.9 kg/b).

No-IHC		Selective	-IHC	Isomerizatio	Isomerization-IHC		
Feed (rank)	(kg CO ₂ /b)	Feed (rank)	(kg CO ₂ /b)	Feed (rank)	(kg CO ₂ /b)		
Salmon oil	62.8	Salmon oil	76.4	Salmon oil	81.0		
Anchovy oil	62.5	Anchovy oil	75.3	Anchovy oil	80.8		
Tuna oil	61.7	Tuna oil	74.8	Tuna oil	80.1		
Menhaden oil	60.7	Soybean oil	72.0	Menhaden oil	78.8		
Soybean oil	55.8	Menhaden oil	71.6	Soybean oil	73.9		
Corn oil–DCO	54.8	Corn oil-DCO	70.6	Corn oil-DCO	72.8		
UCO 90 th P.	54.4	UCO 90 th P.	70.4	UCO 90 th P.	72.6		
Herring oil	54.0	Canola oil	70.3	Herring oil	72.0		
Cottonseed oil	53.4	Cottonseed oil	67.2	Cottonseed oil	71.5		
Canola oil	53.1	Herring oil	66.4	Canola oil	71.1		
Poultry fat	49.4	UCO 10 th P.	62.5	Poultry fat	67.2		
UCO 10 th P.	49.2	Poultry fat	61.7	UCO 10 th P.	67.1		
Lard fat	47.6	Lard fat	60.2	Tallow fat	65.7		
Palm oil	47.1	Tallow fat	60.0	Lard fat	65.3		
Tallow fat	46.2	Palm oil	56.9	Palm oil	64.8		

Table 5. Hydrogen steam reforming emissions from the No-IHC, Selective-IHC, and Isomerization IHC refining strategies: comparisons of whole HEFA feed ranks by emission rate.

IHC: Intentional hydrocracking. **No-IHC:** CO_2 from hydrodeoxygenation (HDO). **Selective-IHC:** CO_2 from HDO plus IHC of HDO output hydrocarbons > C16. **Isomerization-IHC:** CO_2 from HDO plus IHC of all HDO output (> C8). **Menhaden:** a fish. **UCO:** used cooking oil, 10th, 90th percentiles shown. **DCO:** distillers corn oil. Figures shown exclude emissions associated with H₂ losses, depropanation, and inadvertent cracking. Data from Table 3 at 9.82 g CO2/g H2 steam reforming.

3.3.2 Feed chemistry effects on feed rankings for CI

Feedstock CI rankings differ between No-IHC and Selective-IHC processing (Table 5). This is a feed quality impact driven primarily by the different volumes of non-jet fractions sent to IHC among feeds. It boosts the CI of soybean oil from 4.9 kg/b below to 0.4 kg/b above the CI of menhaden oil with the addition of Selective-IHC (*Id.*). With 88.3% of its volume outside the jet fuel range compared with 57.7% of menhaden oil (Table 1, > C16 fractions), soy oil sends 30.6% more feed to Selective-IHC than menhaden oil. More IHC feed requires more hydrogen, boosting steam reforming emissions more with soy than with menhaden oil. Similarly, canola oil sends 27.9% more feed to Selective-IHC than herring oil (*Id.*). This boosts canola oil CI from 0.9 kg/b below to 3.9 kg/b above herring oil CI with the addition of Selective-IHC (Table 5).

3.3.3 How livestock fat feeds could affect soy oil and canola oil refining CI

When switching from soy or canola oil to livestock fat enables a refiner to boost jet fuel yield by repurposing its refinery for Selective-IHC processing, that intentional hydrocracking can boost jet yield from soy and canola oil feeds as well. Thus, instead of shutting down when, for any reason at any time, livestock fat becomes too scarce or expensive, the refiner could make jet fuel by going back to soybean oil or canola oil feedstock. This could increase refining CI by 16.2 kg/b (29%) for soy oil, and 17.2 kg/b (32%) for canola oil, based on our results for the Selective-IHC *versus* No-IHC processing strategies in Table 5.

3.4 Feedstock effects on CI resulting from Isom-IHC hydrogen demand

Livestock fat and palm oil could maximize jet fuel yield by enabling Isom-IHC processing, since these feeds minimize HDO hydrogen demand (chapters 1 and 2). Their relatively lower non-jet fractions do not contribute to this effect on Isom-IHC because, in contrast to Selective-IHC, Isom-IHC processes the entire feed stream output from HDO. Direct effects of feed quality variability on Isom-IHC cracking are relatively weak, since HDO both saturates and removes oxygen from Isom-IHC inputs. Thus, the relative feed rankings for CI from No-IHC processing carry over to the Isom-IHC feed rankings with only minor differences (Table 5). However, by cracking of the entire HDO output, Isom-IHC further boosts hydrogen demand, thus hydrogen steam reforming emissions, resulting in the highest HEFA refining CI for all feeds (*Id.*).

Across feeds and process options, from the lowest to the highest impact combinations of feeds and processing, HEFA refining CI increases by 34.8 kg CO₂/b (75%), and CI increases in 122 (79.7%) of 153 feed switching combinations that could boost jet fuel yield (tables 1, 3, 5).

3.5 <u>Comparison with petroleum refining CI by feedstock and processing strategy</u>

Chart 1 plots results for feedstock-related impacts on the variability of HEFA refining CI from HEFA steam reforming emissions against the CI of U.S. petroleum refining. Our results in Table 5 are shown by processing strategy and, within each strategy, each feed is represented by a color-coded column. The height of the column represents the contribution of steam reforming to HEFA refining CI for that particular feed and processing strategy. The solid black line shown at approximately 41.8 kg/b (kg CO₂/barrel crude processed) represents the average U.S. petroleum refining CI from 2015 through 2017.⁶ We use this (41.8 kg/b) as our benchmark. For added context, average U.S. petroleum refining CI from 2006–2008,⁷ a period when the U.S. refinery crude slate was denser and higher in sulfur than during 2015–2017⁸ resulting in higher historic U.S. crude refining industry CI,⁷ is represented by the dashed line at 50 kg/b in the chart.

Please note what HEFA emissions Chart 1 does and does not show. It shows HEFA refining steam reforming emissions only. This helps us focus on our question about refining CI impacts from HEFA feedstock switching to target jet fuel, which are directly related to HEFA steam reforming rates. It *does not* show total direct emissions from HEFA refining.

3.5.1 HEFA refining CI impacts are significant compared with crude refining

Other HEFA refining emissions besides those from steam reforming—from fuel combustion to heat and pressurize HEFA hydro-conversion reactors, precondition and pump their feeds, and distill and blend their products—could add roughly 21 kg/b of additional HEFA refining CI.² Thus, for a rough comparison of petroleum refining CI with total HEFA refining CI, imagine adding 21 kg/b to the top of each column in Chart 1. HEFA refining CI approaches or exceeds *double* the CI of petroleum refining. Clearly, expanding HEFA jet fuel would increase the CI of hydrocarbon fuels processing substantially.



1. HEFA Steam Reforming Emissions v. Total U.S. Petroleum Refining Emissions, kg CO₂/barrel feed input.

a. HEFA steam reforming emissions only: values shown exclude CO_2 emitted by other HEFA refining process and support equipment. This contrasts with the petroleum refining emissions shown, which include all direct emissions from crude refining. Including all direct emissions from HEFA refining could increase the HEFA estimates shown by approximately 21 kg/barrel.² The "No-IHC" strategy excludes intentional hydrocracking (IHC); the "Selective-IHC" strategy adds emission from producing hydrogen consumed by intentional hydrocracking of feed fractions comprised of hydrocarbons outside the jet fuel range; the "Isomerization-IHC" strategy adds emissions from intentional hydrocracking of whole feeds in the isomerization step of HEFA fuels production. HEFA data shown include feed-driven emissions in Table 5 plus additional steam reforming emissions (2.5 kg/b) from producing the additional hydrogen that is lost to unintended side-reaction cracking, solubilization, scrubbing and purging (*see* Table 8).¹

b. U.S. petroleum refinery emissions including total direct CO₂ emitted from steam reforming and all other petroleum refinery process and support equipment at U.S. refineries. Mean from 2015 through 2017 based on total refinery emissions and distillation inputs reported by the U.S. Energy Information Administration (EIA).⁶ Mean from 2006 through 2008 represents a period of historically high-carbon U.S. refining industry crude inputs.⁷⁸

3.5.2 High-jet feed impacts on processing targeting jet fuel can increase refining CI

Feeds that enable intentional hydrocracking to boost jet fuel yield could increase HEFA refining CI significantly (Chart 1). Here we report feed switching CI increments compared with No-IHC processing of soy and canola oils to target diesel yield (*see* Table 5) as percentages of our petroleum crude refining benchmark: Switching to Selective IHC with anchovy and salmon oils increases CI by 47% to 56% (of crude refining CI) while switching to Selective IHC with menhaden oil increases CI by 38% to 44%. Switching to Isom-IHC with tallow increases CI by 24% to 30% while switching to Isom-IHC with palm oil increases HEFA refining CI by 21% to 28% of crude refining CI. Switching to Selective-IHC with tallow increases CI by 10% to 17%. Only Selective-IHC with palm oil has similar CI to that of No-IHC with soy oil (+3%).

3.5.3 High-jet feed CI impacts are mixed in processing targeting HEFA diesel yield

Compared with No-IHC processing of soy or canola oils, which are the combinations of processing and feeds that maximize HEFA diesel yield, No-IHC with fish oils could increase refining CI while No-IHC with palm oil or livestock fats could decrease CI. For example,

switching to anchovy oil could increase No-IHC HEFA refining CI over that of canola and soy oils by 16% to 23% of crude refining CI while switching to tallow could decrease it by 16% to 23% of crude refining CI. But there is a caveat to those estimates.

In theory, feeding tallow to No-IHC processing could boost jet fuel yield to one-third of feedstock volume (Table 1) while lowering CI by 6.8 or 9.5 kg/b below canola or soy oil in No-IHC processing, the strategies refiners use to maximize HEFA diesel yield. However, this would require three barrels of tallow feed per barrel of jet fuel yield, emphasizing a crucial assumption about HEFA biofuel as a sustainable jet fuel solution—it assumes a sustainable feedstock supply. That assumption could prove dangerously wrong, as shown in Chapter 4.

4. HEFA jet fuel feedstock and carbon sinks: Could the feedstocks that maximize HEFA jet fuel instead of diesel yield have comparatively high indirect climate impacts?

Increasing demand for limited supplies of feedstocks that refiners could use to boost HEFA jet fuel yield and make more HEFA jet fuel risks increasing deforestation and other serious indirect climate impacts. HEFA biofuel feedstocks are purpose-derived lipids also needed for food and other uses,^{9 10} are globally traded, and can increase in price with increased biofuel demand for their limited supply.² Ecological degradation caused by expanded production and harvesting of the extra lipids for biofuels has, in documented cases, led to emissions from natural carbon sinks due to biofuels. Those emissions have traditionally been labeled as an "indirect land use impact," but as shown above, refiners seeking to maximize HEFA jet fuel production also could use fish oil feedstocks. The term "indirect carbon impacts," meant to encompass risks to both terrestrial and aquatic carbon sinks, is used in this chapter.

4.1 Natural carbon sinks that HEFA jet fuel feedstock acquisition could affect

Feedstocks that increase HEFA jet fuel production could have indirect impacts on landbased carbon sinks, aquatic carbon sinks, or both. At the same time the impact mechanisms differ between terrestrial and aquatic ecosystems. Part 4.1.1 below discusses carbon sink risks due to land degradation, and part 4.1.2 discusses carbon sink risks due to fishery depletion.

4.1.1 Land degradation risks: Carbon sinks in healthy soils and forests

Even before new Sustainable Aviation Fuel plans raised the potential for further expansion of HEFA feedstock acquisition, biofuel demand for land-based lipids production was shown to cause indirect carbon impacts. A mechanism for these impacts was shown to be global land use change linked to prices of commodities tapped for both food and fuel.¹¹ Instead of cutting carbon emissions, increased use of some biofuel feedstocks could boost crop prices, driving crop and pasture expansion into grasslands and forests, and thereby degrading natural carbon sinks to result in biofuel emissions which could exceed those of petroleum fuels.¹¹

Indirect carbon impacts of lipid feedstocks which further HEFA biofuel expansion could tap have been observed and documented in specific cases. International price dynamics involving palm oil, soybean oil, biofuels and food were linked as factors in the deforestation of Southeast Asia for palm oil plantations.¹² Soy oil prices were linked to deforestation of the Amazon and Pantanal in Brazil for soybean plantations.^{13 14 15} Demand-driven changes in European and U.S. prices were shown to act across the oil crop and animal fat feedstocks for HEFA biofuels.¹⁶ Rapeseed (canola) and soy biofuels demand drove palm oil expansion in the Global South as palm oil imports increased for other uses of those oils displaced by biofuels in the Global North.¹⁷ Indirect land use impacts of some soy oil—and most notably, palm oil—biofuels were found to result in those biofuels emitting more carbon than petroleum fuels they are meant to replace.^{17 18 19} Current U.S. policy discourages palm oil-derived biofuel for this reason.²⁰

As of 2021, aerial measurements suggest that combined effects of deforestation and climate disruption have turned the southeast of the great Amazonian carbon sink into a carbon source.²¹ Market data suggest that plans for further HEFA biofuels expansion have spurred an increase in soybean and tallow futures prices.^{22 23 24} A joint report by two United Nations-sponsored bodies, the Intergovernmental Panel on Climate Change and the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services, warns that expansion of industrial biofuel feedstock plantations risks inter-linked biodiversity and climate impacts.²⁵

Moreover, these risks are mutually reinforcing. Potential pollinator declines,²⁶ climate heating-driven crop losses,²⁷ biofuel policy-driven food insecurity,²⁸ and the prospect that, once a biofuel also needed for food is locked into place, retroactive limits on land use conversion could worsen food insecurity,¹¹ reveal another aspect of this carbon sink risk. Namely, the assumption asserted by HEFA biofuel proponents, that we can "grow our way out" of limits on biomass diversion to biofuels by increasing crop yields and reverse course later if that does not work, risks lasting harm.

4.1.2 Fishery depletion risks: The biological carbon pump in world oceans

Increasing demand for fish products could further drive fisheries depletion, thereby risking substantial emissions from the oceanic carbon sink. This potential impact, like that on terrestrial carbon sinks, has received intensifying scientific attention in recent years, but appears to remain less widely known to the general public. Fished species have crucial roles in the mechanisms that send carbon into the oceanic carbon sink, as shown below.

Oceans account for 71% of the Earth surface²⁹ and remove roughly one-fourth to one-third of total carbon emissions from all human activities annually.^{30 31} A portion of the CO₂ exchange between air and water at the sea surface is sequestered in the deep seas via inter-linked shallow, mid-reach, and benthic ecosystems that comprise a "biological pump" in which fished species play key roles. *See* Illustration 1.



Illustration 1. Biological pump to the deep oceans carbon sink

Fish have key roles in the inter-linked shallow, mid-reach, and benthic ecosystems that drive a "biological pump" which sends carbon into the deep seas. In well-lit shallow waters, photosynthesis converts CO₂ into organic carbon that is taken up by plants, then by animals in aquatic food webs, and horizontal migration of faster-swimming species fertilizes phytoplankton blooms in the nutrient-poor open oceans, reinforcing the carbon uptake. Some of this carbon falls to the deep sea in fecal pellets and carcasses of fish and other animals (dashed lines shown), while respiration releases CO₂ from aquatic animals and from bacterial degradation of fecal matter (upward-curving lines), some of which re-enters the atmosphere at the sea surface. Active vertical migration (solid vertical lines) further drives the biological pump. A substantial portion of both fish and their invertebrate prey biomass feeds near the surface at night and in much deeper mid-reaches of the ocean during daylight—where deep-sea fish species migrate and feed as well (black and red boxes). Here in the mid-reaches, a greater portion of the carbon in fecal pellets and dead fish sinks to the bottom, and active migration feeding by deep sea fish transfers additional carbon to the deep sea. The organic carbon that reaches the deep sea can be sequestered in sediments for hundreds to thousands of years.

In well-lit shallow waters, photosynthesis converts CO_2 into organic carbon that is taken up by plants and then by animals in ocean food webs. (Illustration, top.) Horizontal migration of faster-swimming species fertilizes phytoplankton blooms in the nutrient-poor open oceans, reinforcing the carbon uptake (*Id.*).^{25 31} Some of this carbon sinks to the deep sea in fecal pellets and carcasses of fish and other animals (dashed lines shown)^{25 32} but not all of it; some of the CO_2 released in respiration by aquatic animals and bacterial degradation of fecal matter re-enters the atmosphere at the sea surface (upward-curving lines).^{30 32} That sea surface carbon exchange emphasizes the role of active vertical migration (solid vertical lines) in the biological pump.

For both fish and their invertebrate prey, a substantial portion of their ocean biomass feeds near the surface at night and in much deeper mid-reaches of the ocean during daylight²⁵—where deep-sea fish species migrate and feed as well.³² Here in the mid-reaches, a greater portion of the carbon in fecal pellets and dead fish sinks to the bottom, and active migration feeding by

deep sea fish transfers additional carbon to the deep sea.^{25 30 32} The organic carbon that reaches the deep sea can be sequestered in sediments for hundreds to thousands of years.^{25 30 32}

Although impacts are not yet fully quantified,²⁵ at present—even at "maximum sustainable yield"—fishery depletion impacts the oceanic carbon sink by removing roughly half of the fisheries biomass that would otherwise be in world oceans.^{25 31} This exports the carbon in fish from ocean sequestration to land, where that exported carbon then enters the atmosphere.^{25 31} Fished species are targeted selectively, disrupting ecosystems involved in the biological pump and potentially reducing both the passive and the active transport of carbon to deep sea carbon sequestration.^{25 32} Worse, as demands for limited fisheries catches have grown, bottom trawling, which directly disrupts and releases carbon from ocean sediments, may already have reduced the oceanic carbon sink by as much as 15–20%.²⁵ In this context fish oil demand, while only a small fraction of total fisheries catch, is still supplied more from whole fish than from fish byproducts, and is projected to grow by a few percentage points through 2030.¹⁰ Thus, potential additional fish oil demand for biofuel poses an indirect carbon impact risk.

4.2 Historic impact assessments for high jet fuel yield HEFA feedstocks

HEFA refiners could maximize jet fuel instead of diesel production using palm oil, fish oil, or livestock fats for feedstocks, as shown in Chapter 2 above. Historic demand for these specific feedstocks has resulted in relatively high indirect carbon impacts from one of them, and raises questions about future impacts from increased demand for the other two high jet fuel yield feeds.

4.2.1 Palm oil: High jet fuel yield, high impact and current use restriction

With 46.5% of its fatty acid feedstock volume comprised of carbon chains in the jet fuel range, palm oil ranks first among major HEFA feedstocks for the potential to increase HEFA jet fuel production. <u>See</u> Table 1. Palm oil also has perhaps the highest known potential among HEFA feedstocks for indirect land use impacts on natural carbon sinks (§ 4.1.1). Some palm oil-derived biofuels have reported fuel chain carbon intensities that exceed those of the petroleum fuels they are meant to replace (*Id*.). However, current U.S. policy restricts the use of palm oil-derived biofuels to generate carbon credits due in large part to this high indirect carbon impact.²⁰ Future biofuel demand could affect the efficacy of this use restriction.

4.2.2 Fish oil: High jet fuel yield and low carbon impact assumed for residual supply

Fish oils rank second, fourth, sixth, seventh and ninth for jet fuel-range fractions at 42.3%, 32.7%, 32.6% and 27.5% of their feed volumes. <u>See</u> Table 1. Moreover, their relatively low diesel fractions (48.9–59.8%) and relatively high feed fractions with carbon chains longer than the ideal diesel range, which could be broken into twin jet fuel hydrocarbons (*Id.*), might favor jet fuel production by intentional hydrocracking strategies. Current biofuel use of fish oil is low, and is assumed to be residual biomass, and thus to have relatively low indirect carbon impact. However, that assumption is based on historic fish oil usage patterns at historic biofuel demand. If HEFA refiners seek to maximize jet fuel production by tapping fish oil in larger amounts, this

has a potential to result in high indirect carbon sink risk by further depleting fisheries that contribute to the biological pump which sequesters carbon in the deep sea (§ 4.1.2).

4.2.3 Livestock fat: High jet fuel yield and low carbon impact assumed for residual supply

Tallow, poultry fat, and lard rank third, fifth, and eighth for jet fuel-range fractions at 33.3%, 32.7%, and 30% of their feed volumes, respectively. <u>See</u> Table 1. For these livestock fats, HEFA feedstock acquisition impact and supply estimates are linked by the assumption that only "waste" residues of livestock fat biomass will be used for biofuels.^{33 34} This results in lower estimates for feedstock acquisition impacts by assuming that impacts from using farm and pastureland to feed the livestock are assigned to other uses of the livestock, such as food. At the same time, this assumption limits the supply for biofuels to only "waste" which, it is assumed, will not result in using more land for livestock feed in response to increased HEFA feedstock demand. These current assumptions—that increased demand will not cause land use impacts because it will not increase livestock production—limit current estimates of both supply and indirect carbon impact. Again, however, the current assumptions driving indirect carbon impact estimates are based on historic lipids usage patterns, which may change with increasing HEFA feedstock demand.

4.3 <u>Feedstock acquisition risks to carbon sinks could be substantial at usage volumes</u> <u>approaching the current HEFA jet fuel blend limit</u>

Impacts of these differences among feedstocks—and HEFA feedstock acquisition impacts overall—depend in large part upon future HEFA demand for limited current feedstock supplies. Moreover, indirect carbon impacts can include impacts associated with displacing other needs for these lipid sources, notably to feed humans directly and to feed livestock or aquaculture fish. This section compares potential HEFA SAF feedstock demand with limited current lipid supplies to assess potential indirect carbon impacts of specific and combined HEFA feedstocks.

4.3.1 Potential future HEFA jet fuel feedstock demand in the U.S.

SAF implementation could drive dramatic HEFA feedstock demand growth. In 2019, the most recent year before COVID-19 disrupted air travel, U.S. SAF consumption was estimated at 57,000 barrels,³⁵ only 0.009% of the 636 million barrels/year (MM b/y) U.S. jet fuel demand.³⁶ Since SAF must be blended with petroleum jet fuel and can be a maximum of half the total jet fuel,³⁵ implementation of SAF goals could result in future jet biofuel production of as much as 318 MM b/y assuming no growth in jet fuel demand. This would represent SAF growth to approximately 5,580 *times* the 2019 SAF biomass demand. HEFA technology is on track to claim the major share of this prospective new biomass demand.

Since 2011, "renewable" diesel production used in California alone, a surrogate for U.S. HEFA biofuel use,³⁵ grew by a factor of 65 times to 2.79 MM b/y as of 2013, by 142 times to 6.09 MM b/y as of 2016, and 244 times to 10.5 MM b/y as of the end of 2019.³⁷ Planned new HEFA capacity targeting the California fuels market and planned for production by 2025 totals approximately 124 MM b/y,³⁸ another potential increase of more than tenfold from 2019–2025.

Financial incentives for oil companies to protect their otherwise stranded refining assets are a major driver of HEFA growth—for example, in the two biggest biorefineries to be proposed or built worldwide to date.² More crude refining asset losses can thus spur more HEFA growth.²

Further idling of crude refining assets is indeed likely. Climate constraints drive the need to replace gasoline, with most credible expert assessments showing approximately 90% of gasoline to be replaced in mid-century climate stabilization scenarios.^{39 40 41 42} More efficient electric vehicles with lower total ownership costs will force gasoline replacement as vehicle stock rolls over, and this independent driver could replace approximately 80% of U.S. gasoline vehicles by mid-century.² Designed and built to co-produce gasoline and maximize gasoline production, U.S. crude refineries cannot produce distillates alone and will be idled as gasoline is replaced.²

Refiners can—and would be highly incentivized to—protect those otherwise stranded assets by repurposing their crude refining equipment for HEFA biofuel production. Assuming the low end of the mid-century crude refining asset loss projections noted above, 80% of existing U.S. refinery hydrogen production capacity could be repurposed to supply approximately 2.66 million metric tons per year (MM t/y) of hydrogen for HEFA production at idled and repurposed crude refineries. <u>See</u> Table 6 below.

Depending on the mix of HEFA jet fuel processing strategies that the prospective new HEFA refining fleet might employ, this much repurposed hydro-conversion capacity could make enough HEFA jet fuel to replace 36% to 39% of total U.S. jet fuel demand, assuming no growth from 2019 demand. *Id.* Notably, if the existing³⁷ and planned³⁸ capacity through 2025 is built and tooled for the same jet fuel yields, this mid-century projection implies a threefold HEFA capacity growth rate from 2026–2050, slower than the tenfold growth planned from 2019–2025.

In order to "book-end" an uncertainty previewed in chapters 1 and 2 above, Table 6 shows two potential HEFA jet fuel growth scenarios. Scenario S-1 assumes a future U.S. HEFA refining fleet with 30% of refineries using the No-IHC strategy and 70% using the Isom-IHC strategy. This scenario assumes many refiners that repurpose for HEFA production lack existing equipment to repurpose for intentional hydrocracking separately and in addition to the hydro-deoxygenation and isomerization reactors needed for all HEFA processing, and refiners choose not to build new hydrocracking capacity into their asset repurposing projects. Scenario S-2 assumes the opposite: many refiners have that existing capacity or choose to build new capacity into their repurposing projects, resulting in a mix with 20% of refineries using the No-IHC strategy, 70% using the Selective-IHC strategy, and 10% using the Isom-IHC strategy.

Relying mainly on Selective-IHC, which cuts hydrogen demand compared with Isom-IHC, Scenario S-2 makes more jet fuel from the same amount of repurposed hydrogen capacity, but nevertheless, at 71–72 MM t/y, feedstock demand is very high in both scenarios (Table 6).

Table 6. Potential HEFA jet fuel growth scenarios to mid-century in the U.S.

t: metric ton	MM t/y: million metric tons/year
---------------	----------------------------------

Total U.S. crude refining Assumption by 2050: 80		3.32 2.66					
Scenario S-1: No use of selective and intentional hydrocracking (Selective-IHC) ^a							
Process strategy		No-IHC	Selective-IHC	Isom-IHC	Total		
Refineries breakdown	(% feed)	30 %	0 %	70 %	100 %		
Hydrogen input ^b	(kg/t feed)	9.04	0.00	28.5	37.5		
Feed input ^b	(MM t/y)	21.3	0.00	49.7	71.0		
Jet fuel yield ^c	(MM t/y)	4.75	0.00	24.5	29.3		
HEFA jet fuel production in the U.S. as a percentage of total 2019 U.S. jet fuel demand:							
Scenario S-2: High use	e of selective a	ind intentional h	ydrocracking (Selec	tive-IHC) ª			
Process strategy		No-IHC	Selective-IHC	Isom-IHC	Total		
Refineries breakdown	(% feed)	20 %	70 %	10 %	100 %		
Hydrogen input ^b	(kg/t feed)	6.02	26.6	4.06	36.7		
Feed input ^b	(MM t/y)	14.5	50.7	7.25	72.4		
Jet fuel yield ^c	Jet fuel yield ° (MM t/y) 3.23 25.0 3.58 31.8						
HEFA jet fuel production in the U.S. as a percentage of total 2019 U.S. jet fuel demand: 39%							

Absent policy intervention, given renewable incentives and assuming severe feed supply limitations are overcome, U.S. HEFA jet fuel production could replace 36–39% of current U.S. petroleum jet fuel, and demand 71–72 million tons/year of lipids feedstock annually, by mid-century. Crude refiners could be highly incentivized to repurpose assets, which would be stranded by climate constraints and electric vehicles, for HEFA biofuels; less clear is the mix of processing strategies the repurposed HEFA refining fleet would use. Refiners could boost jet fuel yield by intentional hydrocracking of HEFA isomerization feeds (Isom-IHC), or do so while limiting hydrogen costs by intentional hydrocracking of selected feed fractions separately from the isomerization step needed for all fractions (Selective-IHC). However, some refineries lack existing equipment for one or both IHC options and may not choose to build onto repurposed equipment. Scenarios in this table span a conservatively wide range of fleet-wide processing strategies in order to "book-end" this uncertainty, resulting in the feed and fuel ranges shown above. The 80% petroleum capacity idling assumed by 2050² is generally consistent with highly credible techno-economic analyses, which, however, generally assume a different biofuel technology and feedstock source.^{40–42} **a.** U.S. refinery hydrogen capacity from *Oil & Gas Journal.*⁵ **b.** Hydrogen and feed inputs based on feed-weighted data from Table 3 and a feed blend SG of 0.914. **c.** Jet fuel yields based on yield-wtd. data from Table 1 at 0.775/0.914 jet/feed SG (No-IHC) and Pearlson et al. (IHC).³ U.S. jet fuel demand in 2019 from USEIA (636.34 MM bbl),³⁶ or 81.34 MM t/y at the petroleum jet fuel density in the survey reported by Edwards (0.804 SG).⁴³ Diesel is the major HEFA jet fuel coproduct. Figures shown may not add due to rounding.

4.3.2 Limited HEFA jet fuel feedstock supplies in the U.S. and world

Current feedstock supplies limit the sustainability of HEFA jet fuel as a substantial component of U.S. jet fuel at rates well below the 50% SAF blend limit. Total current U.S. lipids production for all uses could supply only 29% of the feedstock needed for HEFA jet fuel to replace 36% to 39% of 2019 U.S. jet fuel use, as shown for scenarios S-1 and S-2 in Table 7 below. Other uses of these lipids crucially involve direct and indirect human needs for food, and in these scenarios, U.S. HEFA biofuel alone displaces one-third of all other existing lipids usage globally (Table 7).

Further, at even half the HEFA jet fuel production rates shown in Table 7, current global production of no one lipid source can supply the increased biofuel feedstock demand without displacing significant food system resources. This observation reveals the potential for impacts that cut across multiple prospective HEFA feedstock sources.

Table 7. HEFA feedstock demand in potential U.S. petroleum jet fuel replacement scenarios compared with total current U.S. and world production for all uses of lipids.

U.S. Feedstock No 100% Replacement		36% Scen	36% Scenario S-1		39% Scenario S-2		
Demand Scenarios ^a	NA: blend limit		71.0 M	71.0 MM t/y		72.4 MM t/y	
Current Feed-	U.S.	World	Supply/De	Supply/Demand (%)		nand (%)	
stock Supply	(MM t/y)	(MM t/y)	U.S.	World	U.S.	World	
Palm oil ^b	0.00	70.74	0%	99%	0%	98%	
Fish oil ^c	0.13	1.00	0.18%	1.4%	0.18%	1.4%	
Livestock fat d	4.95	14.16	7%	20%	7%	20%	
Soybean oil ^e	10.69	55.62	15%	78%	15%	77%	
Other oil crops ^e	5.00	73.07	7%	103%	7%	101%	
Total Supply	20.77	214.59	29%	309%	29%	302%	

Compared with total current U.S. and world produc

Total current U.S. production for all uses of lipids also tapped for biofuel could supply only 29% of potential U.S. HEFA jet fuel feedstock demand in 2050. **a.** HEFA feedstock demand data from Table 6. **b.** Palm oil data from Oct 2016–Sep 2020.⁴⁴ **c.** Fish oil data from 2009–2019 (U.S.)⁴⁵ and unspecified recent years (world).⁴⁶ **d.** Livestock fat data from various dates (US)⁹ and 2018 (world).⁴⁷ **e.** Soybean oil, palm oil, and other oil crops data from unspecified dates for used cooking oil (US),⁹ Oct 2016–Sep 2020 for oil crops also used for biofuel (US),⁴⁸ and Oct 2016–Sep 2020 for oilseed crops (world).⁴⁴

4.3.3 Feed-specific and total feed-blend indirect carbon impact potentials

As shown in Table 7 and discussed above, the scale of potential HEFA feedstock demand affects the answer to our question about whether feedstocks refiners could use to increase HEFA jet fuel yield could result in relatively more serious indirect carbon impacts.

Palm oil: High volume displacement and international fueling impacts potential

With the highest global availability of any current HEFA feed (Table 7), palm oil is likely to fill in for current uses of other HEFA feeds that growing U.S. feedstock demand for HEFA jet fuel would displace from those uses. This could occur regardless of restrictions on palm oil biofuel, increasing the indirect carbon impacts associated with palm oil expansion. Deforestation in Southeast Asia caused by palm oil expansion has been linked to biofuel demand for soy and rapeseed (canola) oils in the U.S. and Europe at past, much lower, biofuel feedstock demand, as described in section 4.1.1. Its high global availability also increases the likelihood that, despite U.S. policy, palm oil derived HEFA jet fuel could burn in many commercial flights. Jets may fuel this palm biofuel in various nations—including fueling for the return legs of international flights originating in the U.S. Palm oil can thus be considered a high jet fuel yield and relatively high indirect carbon impact HEFA feedstock.

Fish oil: Unique risk at low HEFA feed blend volume

In contrast to palm oil, fish oil is an extremely low availability HEFA feedstock and is unique among HEFA feeds in raising risks to the oceanic carbon sink. Equally important, fish oil has hard-to-replace aquaculture and pharmaceutical uses.¹⁰ At 1.4% of current world supply for HEFA jet fuel demand scenarios in Table 7, fish oil is unlikely to be targeted as a major

HEFA feedstock industry wide. But this also means that existing uses of fish oil that are hard to replace could be fully displaced, driving further fisheries depletion, even if fish oil comprises as little as 1.4% of potential future HEFA feeds. Increased fishing pressure for fish oil is difficult to discount in demand scenarios approaching those shown (*Id.*), as significant upward pressure on lipids prices could impact lipids markets globally. Indeed, world fish oil demand for all uses is projected to grow and continue to be produced in substantial part from whole fish catch.¹⁰ That fish biomass would essentially be extracted from the oceanic carbon sink to emit carbon from land-based uses, however, the larger and more uncertain impact could be on the effectiveness of ocean carbon sequestration via the biological pump (§ 4.1.2).

Available information thus identifies the potential for a future fish oil biofuel impact which may or may not materialize but nevertheless poses significant risk. Fish oil can be considered a high jet fuel yield and relatively high indirect carbon risk HEFA feedstock.

Livestock fat: likely displacement and possible supply growth impacts

While total current livestock fat production could supply only 20% of potential HEFA feedstock demand (Table 7), its relatively high jet fuel yield and relatively low (assumed) indirect carbon impacts could make livestock fat an important fraction of the expanding HEFA feeds mix. This would displace its existing uses, where the fats would likely be replaced by expanded demand for other lipids with relatively higher indirect carbon impacts. High-availability replacements such as palm and soy oils (*Id.*) would likely fill those displaced uses, and both palm and soy oils have relatively high indirect carbon impacts (§ 4.1.1).

Additionally—and notwithstanding the likelihood that livestock protein production would remain the priority—it is possible that the unprecedented growth in livestock fat demand might alter the balance among choices for producing human protein intake in favor of this high jet fuel yield "byproduct" feedstock. This balance is dynamic, as suggested by trends either toward or away from vegetarian diets in various human populations globally, such that this possibility is difficult to discount given the potential for unprecedented livestock fat demand growth. And if HEFA demand were to drive livestock production growth, livestock production is, in fact, a high carbon emission enterprise.^{31 49} In view of these likely and possible impacts, livestock fat can be considered a high jet fuel yield and relatively high indirect carbon risk HEFA feedstock.

Feed blends: limited residue supply worsens indirect carbon impacts

Impacts and risks of high jet fuel yield feedstock add to those of feed blends that could be used for HEFA jet fuel, and limited global "residue" feedstock supply heightens these impacts.

HEFA feedstock demand to replace just 18% of 2019 U.S. jet fuel use—half that shown in Table 7—would far exceed current total U.S. production for *all uses* of lipids also tapped for biofuels. One implication of this is the need to consider food and fuel uses of the global lipids supply by other nations. Importantly, at 4.28% of world population, the U.S. per capita share of world production for low impact "residue" feeds from livestock fat and fish oil (Table 7) is less

than 0.65 MM t/y, less than 1% of potential U.S. HEFA jet fuel feedstock demand (*Id.*). The limited supply of low impact "residue" feedstocks, in turn, limits alternatives to palm oil or livestock production growth that can feed potential HEFA jet fuel growth. Current major feed alternatives for HEFA jet fuel are limited to soybean oil and other oil crops (*Id.*).

For example, what if U.S. palm biofuel is prohibited, livestock and fish oil production do not grow, and U.S. HEFA "residue" feedstock acquisition grows to eight times its per capita share (5.2 MM t/y)? At half of its minimum potential mid-century growth, HEFA feedstock demand for SAF in the U.S. would be approximately 35.5 MM t/y (Table 7). This 5.2 MM t/y of low-impact feed would meet only 15% of that demand and leave 30.3 MM t/y of that demand unmet. Supplying the 30.3 MM t/y of unmet demand for just half of potential U.S. HEFA jet fuel growth could induce growth of 23.5% in current combined global production for soy and other oil crops, excluding palm oil (*Id.*).

Moreover, the excess U.S. use of limited global residue supply in the example above could have an impact. It could displace the lower-impact HEFA jet fuel feed for SAF fueled in other nations, which could replace residue feeds with higher indirect carbon impact feeds. This would only shift emissions to HEFA jet fueling elsewhere, without providing a global climate benefit.

Thus, even if U.S. policy effectively discourages palm oil biofuel and livestock production does not grow, the potential HEFA jet fuel expansion could be expected to spur an expansion of soybean, corn, and other plant oil crops. Significant indirect carbon impacts have been linked to biofuels demand for soybean and other plant oil feedstocks at past biofuel demand levels that were substantially lower than current and potential future HEFA demand (§ 4.1.1). While this complicates the answer to our question about indirect carbon impacts of feeds to boost HEFA jet fuel yield, importantly, it further informs our answer. It shows that these heightened impacts and risks would add to significant potential impacts of increased total HEFA feedstock demand.

In plausible future SAF implementation scenarios, among the relatively high jet fuel yield feedstocks, palm oil could have relatively serious indirect carbon impacts, and both fish oil and livestock fat could pose relatively serious but currently uncertain indirect carbon impact risks. Those impacts and risks would add to significant potential carbon sink impacts from the blends of feedstocks that could supply HEFA refineries, in which lower impact "residue" feedstocks could supply only a small fraction of total HEFA feedstock growth. Natural limits on total supply for the type of feedstock that HEFA technology can process appear to make replacing any significant portion of current petroleum jet fuel use with this type of biofuel unsustainable.

5. Limitations and suggestions for future work

Two types of data limitations which may affect potential outcomes for SAF were identified in the course of this research. The first involves HEFA technology: interchangeability among other uses of its feedstocks; and its potential future evolution. These HEFA-specific limitations are discussed in Section 5.1 below. The second involves other alternatives to petroleum jet fuel combustion which, though they are outside the scope of this report, warrant mention due to limitations of HEFA technology identified by this research. These are discussed briefly as suggested priorities for future work in Section 5.2.

5.1 <u>HEFA biofuel impact assessment data limitations</u>

5.1.1 Limited cross-feed displacement quantification data

HEFA feedstocks are not "wastes." All of them are lipids, and more specifically, triacylglycerols of fatty acids, which can be converted to functionally similar biological or chemical uses by many biological processes (e.g., digesting food) and chemical processes (e.g., HEFA processing with hydrocracking). Further, these lipids have interchangeable and largely competing uses now, including food for human populations, livestock feeds, pet food, aquaculture feeds, and feedstocks for making soap, wax, lubricants, plastics, natural pigments, cosmetic products and pharmaceutical products.^{9 10} Accordingly, increased biofuel demand for one source of these lipids displaces another existing use of that feedstock, thereby increasing demand and prices for other sources of lipids as well. Indeed, this has occurred, leading to indirect land use impacts that increased carbon emissions associated with biofuels (§ 4.1.1).

For example, if diverting tallow from soap making to HEFA jet fuel forces soap makers to use more palm oil, that jet fuel indirectly emits carbon associated with that extra production of palm oil. The livestock fat biofuel would cause an indirect carbon impact that current biofuel impact accounting practices for "waste" residue feedstocks assume it does not cause.

However, the hypothetical extreme wherein all lipids are 100% fungible, and any increase in HEFA demand for any of these feedstocks would have the same indirect impact by increasing collective demand for all other feeds by the same amount, also seems unrealistic. Some types of lipids, such as those that increase jet fuel production and those people eat directly, could attract relatively higher demand and command relatively higher prices. At present, *how much* demand increase for each lipid source increases indirect carbon impacts associated with cross-feed demand increase has not yet been quantified by universally accepted estimates.

Herein, we take the view that the uses of lipids also tapped for HEFA biofuels are fungible to a significant extent which varies among specific lipids sources and uses. In this view, indirect carbon impacts of future demand for palm oil exceed those of other HEFA feeds which would not be favored by refiners seeking to boost jet fuel production, but by amounts that are not yet fully quantifiable. That quantitative uncertainty results from the data limitations discussed above and explains why this report does not attempt to quantify the feed-specific indirect carbon impacts documented in Chapter 4.

5.1.2 Renewable fuel hydrogen specification error

Splitting water with electricity supplied by solar or wind power—renewable powered electrolysis—produces zero-emission hydrogen fuel. Unfortunately, renewable fuel standards incentivize HEFA fuels even though much of the hydrogen in those hydrocarbons is produced from non-renewable fossil fuels. This is a mistake. This mistake has led to an important limitation in the data for assessing the future potential of HEFA jet fuel.

Hydrogen steam reforming repurposed from crude refining drives the high CI of HEFA refining and its variability among HEFA feedstocks and processing strategies (Chapter 3). Renewable-powered electrolysis could eliminate those steam reforming emissions and result in HEFA refining CI lower than that of petroleum refining.² However, the combination of public incentives to refiners for HEFA biofuel, and their private incentives to avoid costs of stranded steam reforming assets they could repurpose and electrolysis they need not build to reap those public incentives, has resulted in universal reliance on steam reforming in HEFA processing. Would the public incentives outweigh the private incentives and cut refining CI if this mistake were corrected, or would the companies decide that another alternative to HEFA jet fuel is more profitable? Since current fuel standards allow them to maximize profits by avoiding the question, there are no observational data to support either potential outcome.

Additionally, if refiners were to replace their steam reformers with renewable-powered electrolysis, energy transition priorities could make that zero-emission hydrogen more valuable for other uses than for biofuel,² and biomass feed costs also would weigh on their decisions.¹⁹ Thus, for purposes of the potential impacts assessment herein, and in the absence of observational data on this question, we take the view that assuming HEFA refining without steam reforming emissions would be speculative, and would risk significant underestimation of potential HEFA jet fuel impacts.

5.1.3 Proprietary catalyst development data

Catalysts are crucial in HEFA refining, and although many catalyst data are claimed as trade secrets, their refining benefits are typically advertised, especially if new catalysts improve yields. The search for a new catalyst that can withstand the severe conditions in HEFA reactors and improve processing and yields has been intensive since at least 2013.^{50 51 52 53 54 55 56}

From this we can infer two things. First, given the maturity of the hydro-conversion technology crude refiners repurpose for HEFA refining, and that long and intensive search, a newly invented catalyst formulation which improves reported HEFA jet fuel yield significantly appears unlikely. Second, given the incentive, the invention of such a new catalyst is possible. Again, however, many specific catalyst data are not reported publicly. Our findings herein are based on publicly reported, independently verifiable data. This limitation in publicly reported catalysis data thus has the potential to affect our yields analysis.

5.2 Priorities for future work

5.2.1 Cellulose biomass alternatives—what is holding them back?

Cellulosic residue biomass such as cornstalks, currently composted yard cuttings, or sawdust can be used as feedstock by alternative technologies which qualify as SAF.^{19 35} Using this type of feedstock for SAF could lessen or avoid the indirect carbon impacts from excessive HEFA jet fuel demand for limited lipids biomass that are described in Chapter 4. Indeed, economy-wide analyses of the technologies and measures to be deployed over time for climate stabilization suggest prioritizing cellulosic biomass, to the extent that biofuels will be needed in some hard-to-decarbonize sectors.^{42 57 58} Despite its promise, however, the deployment of cellulosic distillate biofuel has stalled compared with HEFA biofuel. Less clear are the key barriers to its growth, the measures needed to overcome those barriers, and whether or not those measures and the growth of cellulosic jet fuel resulting from them could ensure that SAF goals will be met sustainably. This points to a priority for future work.

5.2.2 Alternatives to burning jet fuel-need and potential to limit climate risks

Even complete replacement of petroleum jet fuel with SAF biofuel combustion would result in ongoing aviation emissions, and would thus rely on additional and separate carbon capturesequestration to give us a reasonable chance of stabilizing our climate. At the current jet fuel combustion rate the scale of that reliance on "negative emission" technologies, which remain unproven at that scale, is a risky bet. Meanwhile, besides alternative aircraft propulsion systems, which are still in the development stage, there are alternatives to jet fuel combustion which are technically feasible now and can be used individually or in combination.

Technically feasible alternatives to burning jet fuel include electrified high-speed rail, fuel cell powered freight and shipping to replace air cargo, and conservation measures such as virtual business meetings and conserving personal air-miles-traveled for personal visits. While we should note that such travel pattern changes raise social issues, so does climate disruption, and

most people who will share our future climate are not frequent fliers. Importantly as well, public acceptance of new travel alternatives is linked to experiencing them. Thus, biofuel limitations, climate risks, and human factors suggest needs to prioritize the development and deployment of alternatives to petroleum jet fuel that do not burn carbon.

5.2.3 Limited safety data record for flying with new fuels

Jet biofuels appear to differ from petroleum jet fuels in their cold flow properties at high altitude, combustion properties, and potential to damage fuel system elastomer material.¹⁹ Those that can be used as SAF have been approved subject to blending limits, which permit SAF to be "dropped-in" to conventional jet fuel up to a maximum of 50% of the blend.⁵⁹ All seven types of biofuels approved for SAF are subject to this condition.⁵⁹ SAF/petroleum jet fuel blends that do not meet this condition are deemed to present potential safety issues.⁵⁹

However, remarkably limited historical use of SAF (§4.3.1) has resulted in a limited data record for assessing its safety in actual operation. That is important because new hazards which result in dangerous conditions over long periods of operation have repeatedly been discovered only by rigorous post-operational inspection or post-incident investigation, the histories of both industrial and aviation safety oversight show. There is an ongoing need to ensure flight safety risks of biofuels are closely monitored, rigorously investigated, transparently communicated, and proactively addressed by "inherent safety measures"⁶⁰ designed to eliminate any specific hazards identified by that future work.

Fatty acid (FA) in HEFA oil feed				Oxygen	Carbon	FA-specific	hydrogen inputs
			Density	content	double	Deoxygenation ^c	Saturation ^{d. e}
common name	Shorthand	Formula ^b	(kg/b)*	(wt. %)*	bonds	$(kg H_2/b)$	(kg H ₂ /b)
Caprylic Acid	C8:0	C ₈ H ₁₆ O ₂	145	22.2	0	8.09	0.00
Capric Acid	C10:0	C ₁₀ H ₂₀ O ₂	142	18.6	0	6.65	0.00
Lauric Acid	C12:0	C12H24O2	140	16.0	0	5.63	0.00
Myristic Acid	C14:0	C14H28O2	137	14.0	0	4.84	0.00
Myristoleic Acid	C14:1	C14H26O2	143	14.1	1	5.10	1.27
Pentadecanoic Acid	C15:0	C ₁₅ H ₃₀ O ₂	134	13.2	0	4.45	0.00
Palmitic Acid	C16:0	C16H32O2	135	12.5	0	4.26	0.00
Palmitoleic Acid	C16:1	C ₁₆ H ₃₀ O ₂	142	12.6	1	4.50	1.13
Margaric Acid	C17:0	C17H34O2	136	11.8	0	4.04	0.00
Stearic Acid	C18:0	C ₁₈ H ₃₆ O ₂	134	11.2	0	3.79	0.00
Oleic Acid	C18:1	C18H34O2	141	11.3	1	4.04	1.01
Linoleic Acid	C18:2	C18H32O2	143	11.4	2	4.12	2.06
Linolenic Acid	C18:3	C18H30O2	145	11.5	3	4.21	3.16
Stearidonic Acid	C18:4	C18H28O2	148	11.6	4	4.33	4.33
Arachidic Acid	C20:0	C20H40O2	131	10.2	0	3.38	0.00
Gondoic Acid	C20:1	C20H38O2	140	10.3	1	3.65	0.91
Eicosadienoic Acid	C20:2	C20H36O2	144	10.4	2	3.76	1.88
Homo-y-linoleic Acid	C20:3	C20H34O2	146	10.4	3	3.84	2.88
Arachidonic Acid	C20:4	C20H32O2	147	10.5	4	3.88	3.88
Eicosapentaenoic Acid	C20:5	C20H30O2	150	10.6	5	4.00	5.00
Henicosanoic Acid	C21:0	$C_{21}H_{42}O_2$	142	9.80	0	3.50	0.00
Heneicosapentaenoic Acid	C21:5	$C_{21}H_{32}O_2$	149	10.1	5	3.79	4.74
Behenic Acid	C22:0	$C_{22}H_{44}O_2$	131	9.39	0	3.09	0.00
Erucic Acid	C22:1	C22H42O2	137	9.45	1	3.26	0.81
Docosadienoic Acid	C22:2	$C_{22}H_{40}O_2$	143	9.51	2	3.43	1.71
Docosatetraenoic Acid	C22:4	C22H36O2	151	9.62	4	3.66	3.66
Docosapentaenoic Acid	C22:5	$C_{22}H_{34}O_2$	148	9.68	5	3.62	4.52
Docosahexaenoic Acid	C22:6	$C_{22}H_{32}O_2$	150	9.74	6	3.68	5.52
Lignoceric Acid	C24:0	$C_{24}H_{48}O_2$	140	8.68	0	3.06	0.00
Tetracosenoic Acid	C24:1	C24H46O2	141	8.73	1	3.11	0.78

Table 8. Data and methods table for feed-specific estimates.^a

* b (barrel): 42 U.S. gallons; wt. %: weight percent on fatty acid

a. See notes to this table for feedstock-specific data sources.

b. Formula symbols; carbon: C (12.011 g/mol); hydrogen: H (1.00794 g/mol); oxygen: O (15.995 g/mol).

c. Deoxygenation: Hydrogen consumed to remove and replace oxygen and propane knuckle-fatty acid bonds.

b. Saturation: Hydrogen consumed to saturate carbon double bonds in HEFA processing.

e. Additional process hydrogen consumption in side-reaction cracking, solubilization, scrubbing and purge losses not shown.

Whole feed fatty acids		Selected plant	oils, livestock	fats and fis	h oils			
Fatty acid	FA	Media	an of sample	analysis pro	file data report	ted based on (C18:2, in wt. %	a
Common name	Shorthand	Soybean	Corn	Canola	Cottonseed	Palm	Tallow	Lard
Caprylic	C8:0					0.186		
Capric	C10:0					0.324		0.070
Lauric	C12:0					2.284	1.010	
Myristic	C14:0	0.100		0.040	0.860	1.108	3.384	1.280
Myristoleic	C14:1							
Pentadecanoic	C15:0							
Palmitic	C16:0	11.000	12.860	4.248	23.600	41.480	24.495	25.000
Palmitoleic	C16:1	0.100	0.100	0.287	0.360	0.167	4.040	3.000
Margaric	C17:0			0.069		0.059	2.020	0.330
Stearic	C18:0	4.000	1.760	1.752	2.400	4.186	17.525	12.540
Oleic	C18:1	23.400	26.950	60.752	17.740	39.706	42.121	44.000
Linoleic	C18:2	53.200	55.880	20.713	54.420	9.902	3.293	11.000
Linolenic	C18:3	7.800	1.260	8.980	0.600	0.196	1.818	0.550
Stearidonic	C18:4							
Arachidic	C20:0	0.300	0.390	0.713	0.220	0.304	0.313	0.190
Gondoic	C20:1		0.280	1.277	0.070	0.078	0.081	0.800
Eicosadienoic	C20:2							0.740
Homo-y-linoleic	C20:3							0.110
Arachidonic	C20:4							0.300
Eicosapentaenoic	C20:5							
Henicosanoic	C21:0							
Heneicosapentaenoic	C21:5							
Behenic	C22:0	0.100	0.120	0.307	0.110	0.039		
Erucic	C22:1			0.594				
Docosadienoic	C22:2							
Docosatetraenoic	C22:4		0.120					
Docosapentaenoic	C22:5		0.180					
Docosahexaenoic	C22:6							
Lignoceric	C24:0			0.099		0.049		
Tetracosenoic	C24:1							
Whole feed FAs	O ₂ wt. %	11.50	11.50	11.35	11.71	11.99	11.80	11.66
Deoxygenatio	on (kg H_2 /b)	4.11	4.11	4.06	4.14	4.19	4.11	4.13
Saturatio	on (kg H_2 /b)	1.58	1.48	1.35	1.32	0.61	0.60	0.76
C8–C16 Fraction	(vol. %)	11.71	13.56	4.78	25.67	46.47	33.34	30.00
Deoxygenatio	on (kg H_2 /b)	4.27	4.26	4.28	4.28	4.38	4.39	4.32
Saturatio	on (kg H_2 /b)	0.01	0.01	0.07	0.02	0.004	0.14	0.12
C15–C18 Fraction	(vol. %)	99.46	98.88	96.85	98.70	95.63	95.18	96.53
Deoxygenatio	on (kg H_2 /b)	4.11	4.11	4.08	4.13	4.13	4.08	4.09
Saturatio	on (kg H_2 /b)	1.59	1.48	1.37	1.34	0.64	0.63	0.75
> C18 Fraction	(VOI. %)	0.43	1.12	3.11	0.42	0.49	0.41	2.10
Deoxygenatio	on (kg H ₂ / b)	3.31	3.49	3.43	3.35	3.37	3.43	3.70
Saturatio	$(\text{Kg} H_2/\text{D})$	0.00	1.38	0.55	0.16	0.15	0.19	1.68

Table 8. Data and methods table for feed-specific estimates continued.^a

Whole feed fatty acids		Selected plant oils, livestock fats and fish oils, continued								
Fatty acid	FA		Median of sar	mple analysis p	rofile data repo	rted based on C1	l8:2, wt. % ^a			
Common name	Shorthand	Poultry	Anchovy	Herring	Menhaden	Salmon	Tuna			
Caprylic	C8:0									
Capric	C10:0									
Lauric	C12:0									
Myristic	C14:0	0.618	6.636	7.755	8.602	6.044	5.903			
Myristoleic	C14:1	0.206					0.447			
Pentadecanoic	C15:0		0.701	0.408	0.538	0.769	0.359			
Palmitic	C16:0	24.206	16.355	15.306	21.505	17.143	17.670			
Palmitoleic	C16:1	6.951	7.757	8.469	10.108	2.198	5.961			
Margaric	C17:0	0.108	0.935	0.510	1.075	1.099	0.650			
Stearic	C18:0	5.814	3.738	2.143	3.333	2.637	4.155			
Oleic	C18:1	42.157	12.150	17.245	15.000	15.385	16.078			
Linoleic	C18:2	18.137	1.636	1.633	2.151	1.648	1.068			
Linolenic	C18:3	0.657	5.607	0.612	1.398	4.451	1.748			
Stearidonic	C18:4		2.336	2.551	3.333	3.077				
Arachidic	C20:0		0.841		0.323	0.385	0.408			
Gondoic	C20:1	0.392	3.738	11.224	1.075	1.978	4.922			
Eicosadienoic	C20:2						0.272			
Homo-y-linoleic	C20:3						3.437			
Arachidonic	C20:4		2.103	0.408	1.720	2.967	0.184			
Eicosapentaenoic	C20:5		14.486	8.776	13.441	12.637	9.282			
Henicosanoic	C21:0									
Heneicosapentaenoic	C21:5		1.869		0.806	2.582				
Behenic	C22:0	0.118					0.078			
Erucic	C22:1	0.098	3.224	15.102	0.645	6.099	0.311			
Docosadienoic	C22:2									
Docosatetraenoic	C22:4									
Docosapentaenoic	C22:5		1.869	1.327	2.258	3.077	5.252			
Docosahexaenoic	C22:6		14.252	6.327	12.366	15.385	20.670			
Lignoceric	C24:0	0.098					0.845			
Tetracosenoic	C24:1	0.363					0.583			
Whole feed FAs	O ₂ wt. %	11.70	11.33	11.22	11.53	11.11	11.20			
Deoxygena	tion (kg H_2 /b)	4.13	4.06	3.99	4.13	4.01	4.01			
Satura	tion (kg H_2 /b)	0.91	2.34	1.52	2.08	2.42	2.31			
C8–C16 Fraction	(vol. %)	32.69	32.56	32.73	42.26	27.48	31.46			
Deoxygena	tion (kg H_2 /b)	4.33	4.45	4.47	4.45	4.42	4.44			
Satura	tion (kg H_2 /b)	0.25	0.28	0.30	0.28	0.09	0.24			
C15-C18 Fraction	(vol. %)	98.09	52.19	49.34	59.81	49.73	48.92			
Deoxygena	tion (kg H_2 /b)	4.13	4.20	4.20	4.21	4.17	4.17			
Satura	tion (kg H ₂ /b)	0.92	1.02	0.89	0.85	1.01	0.64			
	(VOI. %)	2.21	40.93	42.68	31.25	43.96	44.52			
Satura	tion $(kg H_2/b)$	0.67	4 31	2.55	4 83	4.27	4 34			
Deoxygena Satura > C18 Fraction Deoxygena Satura	(vol. %) tion (kg H ₂ /b) tion (kg H ₂ /b) tion (kg H ₂ /b) tion (kg H ₂ /b)	4.13 0.92 1.07 3.31 0.67	4.20 1.02 40.93 3.76 4.31	4.20 0.89 42.68 3.59 2.52	4.21 0.85 31.25 3.81 4.83	4.17 4.17 1.01 43.96 3.72 4.27	4.17 0.64 44.52 3.72 4.34			

Table 8. Data and methods table for feed-specific estimates continued.^a

Whole feed fatty acids		Used cooking oil (UCO) variability						
Fatty acid	FA	Percentiles on C18:2, in wt. % *							
Common name	Shorthand	10 th Percentile	25 th Percentile	75 th Percentile	90 th Percentile				
Caprylic	C8:0								
Capric	C10:0								
Lauric	C12:0								
Myristic	C14:0	0.909	2.479	1.735					
Myristoleic	C14:1								
Pentadecanoic	C15:0								
Palmitic	C16:0	20.606	20.248	16.412	12.420				
Palmitoleic	C16:1	4.646		1.735					
Margaric	C17:0								
Stearic	C18:0	4.848	12.810	5.235	5.760				
Oleic	C18:1	53.434	38.017	29.843	26.930				
Linoleic	C18:2	13.636	23.967	41.324	49.600				
Linolenic	C18:3	0.808	2.066	3.500	4.930				
Stearidonic	C18:4								
Arachidic	C20:0	0.121			0.750				
Gondoic	C20:1	0.848							
Eicosadienoic	C20:2								
Homo-y-linoleic	C20:3								
Arachidonic	C20:4								
Eicosapentaenoic	C20:5								
lenicosanoic	C21:0								
Heneicos apenta enoic	C21:5								
Behenic	C22:0	0.030							
Erucic	C22:1	0.071							
Docosadienoic	C22:2								
Docosatetraenoic	C22:4								
Docosapentaenoic	C22:5								
Docosahexaenoic	C22:6								
Lignoceric	C24:0	0.040							
letracosenoic	C24:1								
Whole feed FAs	O ₂ wt. %	11.64	11.59	11.59	11.55				
Deoxygen	ation (kg H_2 /b)	4.11	4.09	4.12	4.10				
Satur	ation (kg H_2 /b)	0.91	0.95	1.29	1.44				
C8–C16 Fraction	(vol. %)	26.81	23.49	20.61	12.90				
Deoxygen	ation (kg H ₂ /b)	4.32	4.32	4.33	4.26				
Satur	ation (kg H_2 /b)	0.20	0.00	0.10	0.09				
L15-C18 Fraction	(vol. %)	97.95	97.46	98.21	99.19				
Deoxygen	ation (kg H_2 / b)	4.11	4.08	4.11	4.10				
Satur		0.92	0.97	1.31	1.46				
Deoxygen	ation (kg Ha /b)	3.56	0.00	0.00	3 38				
Satur	ation (kg H, /b)	0.75	0.00	0.00	0.00				
Satur		0.75	0.00	0.00	0.00				

Table 8. Data and methods table for feed-specific estimates continued.^a

Table 8. Data and methods table for feed-specific estimates continued.^a

Data for feedstock fractions outside the jet fuel range (> C16)

Feedstock		Soybean	Corn	Canola	Cottonseed	Palm	Tallow	Lard
> C16 Fraction	(vol. %)	88.29	86.44	95.22	74.33	53.53	66.66	70.00
Deoxygenation (kg H ₂ /b)		4.09	4.08	4.05	4.09	4.03	3.98	4.00
Saturation (kg H_2 /b)		1.78	1.70	1.41	1.75	1.12	0.82	1.03

Feedstock		Poultry	Anchovy	Herring	Menhaden	Salmon	Tuna
> C16 Fraction	(vol. %)	67.31	67.44	67.27	57.74	72.52	68.54
Deoxygenati	on (kg H_2 /b)	4.03	3.88	3.76	3.92	3.86	3.82
Saturation (kg H_2 /b)		1.22	3.29	2.10	3.33	3.25	3.21

Feedstock Percentile on C18:2 in wt.%		Used Cooking C	il (UCO		
		10th	25th	75th	90th
> C16 Fraction	(vol. %)	73.19	76.51	79.39	87.10
Deoxygenation (kg H ₂ /b)		4.03	4.03	4.07	4.07
Saturation (kg H_2 /b)		1.16	1.23	1.58	1.65

Table 8. Data and methods table for feed-specific estimates continued.^a

HDO & ONLY	Jet range (C8–C16)			Diese	Diesel range (C15–C18)			Longer chains (> C18)		
(No-IHC)	(vol.%)	Ox (kg/b)	Sat (kg/b)	(vol.%)	Ox (kg/b)	Sat (kg/b)	(vol.%)	Ox (kg/b)	Sat (kg/b)	
High jet/high diesel										
Palm oil	46.47	4.38	0.004	95.63	4.13	0.64	0.49	3.37	0.15	
Tallow fat	33.34	4.39	0.14	95.18	4.08	0.63	0.41	3.43	0.19	
Poultry fat	32.69	4.33	0.25	98.09	4.13	0.92	1.07	3.31	0.67	
Lard fat	30.00	4.32	0.12	96.53	4.09	0.75	2.10	3.70	1.68	
UCO 10th P.	26.81	4.32	0.20	97.95	4.11	0.92	1.12	3.56	0.75	
Cottonseed oil	25.67	4.28	0.02	98.70	4.13	1.34	0.42	3.35	0.16	
High jet/low diesel										
Menhaden oil	42.26	4.45	0.28	59.81	4.21	0.85	31.25	3.81	4.83	
Herring oil	32.73	4.47	0.30	49.34	4.20	0.89	42.68	3.59	2.52	
Anchovy oil	32.56	4.45	0.28	52.19	4.20	1.02	40.93	3.76	4.31	
Tuna oil	31.46	4.44	0.24	48.92	4.17	0.64	44.52	3.72	4.34	
Salmon oil	27.48	4.42	0.09	49.73	4.17	1.01	43.96	3.72	4.27	
Low jet/high diesel										
Corn (DCO) oil	13.56	4.26	0.01	98.88	4.11	1.48	1.12	3.49	1.38	
UCO 90th P.	12.90	4.26	0.09	99.19	4.10	1.46	0.81	3.38	0.00	
Soybean oil	11.71	4.27	0.01	99.46	4.11	1.59	0.43	3.31	0.00	
Canola oil	4.78	4.28	0.07	96.85	4.08	1.37	3.11	3.43	0.55	

Process hydrogen consumption by feedstock and processing strategy (kg/b feed)

HDO & INTENTIONAL	HDO Δ (Ox + Sat)			Intentional Hydrocracking (IHC)			Jet target H2 Δ by processing case		
HYDROCRACKING	Jet rg.	Diesel rg.	> C18	Selecti	ve-IHC	Isom IHC	No-IHC	Select-IHC	Isom-IHC
vol. weighted data	(kg/b)	(kg/b)	(kg/b)	(b fraction)	(kg/b)	(kg/b)	(kg/b)	(kg/b)	(kg/b)
High jet/high diesel	—frac	tions do not	add-	> C16	(factor)*	(factor)*	whole feed	whole feed	whole feed
Palm oil	2.04	4.57	0.02	0.535	1.87	1.80	4.79	5.79	6.60
Tallow fat	1.51	4.47	0.01	0.667	2.10	1.99	4.71	6.11	6.70
Poultry fat	1.50	4.95	0.04	0.673	1.85	1.82	5.03	6.28	6.85
Lard fat	1.33	4.67	0.11	0.700	1.84	1.81	4.85	6.13	6.65
UCO 10th P.	1.21	4.92	0.05	0.732	1.85	1.82	5.01	6.37	6.83
Cottonseed oil	1.10	5.40	0.01	0.743	1.88	1.84	5.44	6.84	7.28
High jet/low diesel									
Menhaden oil	2.00	3.03	2.70	0.577	1.93	1.84	6.18	7.30	8.02
Herring oil	1.56	2.51	2.61	0.673	1.87	1.83	5.50	6.76	7.33
Anchovy oil	1.54	2.72	3.30	0.674	1.93	1.86	6.37	7.67	8.23
Tuna oil	1.47	2.35	3.59	0.685	1.94	1.87	6.29	7.62	8.16
Salmon oil	1.24	2.57	3.51	0.725	1.91	1.85	6.40	7.78	8.25
Low jet/high diesel									
Corn (DCO) oil	0.58	5.53	0.05	0.864	1.86	1.84	5.58	7.19	7.42
UCO 90th P.	0.56	5.51	0.03	0.871	1.87	1.84	5.55	7.17	7.39
Soybean oil	0.50	5.67	0.01	0.883	1.86	1.84	5.68	7.33	7.52
Canola oil	0.21	5.28	0.12	0.952	1.85	1.84	5.40	7.16	7.24

Note: H₂ inputs shown exclude side-reaction cracking, solubilization, scrubbing and purge gas losses.

* IHC $\rm H_2$ consumption at 1.3 wt. % feed (Pearlson et al.), in kg/b IHC input.

See table notes next page

Explanatory notes and data sources for Table 8.

Feeds shown have been processed in the U.S. except for palm oil, which is included because it is affected indirectly by U.S. feedstock demand and could be processed in the future, possibly in the U.S. and more likely for fueling international flights in various nations. Median values shown for feed composition were based on the median of the data cluster centered by the median value for C18:2 (linoleic acid) for each individual whole feed. Blend data were not available for used cooking oil (UCO), except in the form of variability among UCO samples collected, which showed UCO to be uniquely variable in terms of HEFA processing characteristics. The table reports UCO data as percentiles of the UCO sample distribution.

Data for feedstock composition were taken from the following sources:

```
Soybean oil<sup>54 55 61 62 63 64 65 66</sup>
Corn oil (distillers corn oil)<sup>54 61 63 65 67 68 69 70</sup>
Canola oil (includes rapeseed oil)<sup>54 55 61-65 67 69 71 72 73</sup>
Cottonseed oil<sup>54 55 63 65 67</sup>
Palm oil<sup>54 55 62-65 67 68 74</sup>
Tallow (predominantly beef fat)<sup>54 64 69 71 75 76 77 78 79</sup>
Lard (pork fat)<sup>68 76 79</sup>
Poultry fat<sup>54 69 76 79 80</sup>
Anchovy<sup>81</sup>
Herring<sup>82 83</sup>
Menhaden<sup>54 81 82</sup>
Salmon<sup>81 83</sup>
Tuna<sup>81 84 85</sup>
```

```
Used cooking oil (UCO)<sup>74 78 86 87 88 89 90 91 92</sup>
```

Hydrogen consumption to deoxygenate and saturate feeds was calculated from fatty acids composition data for each feed and feed fraction shown. Note that O_2 wt.% data shown are for fatty acids excluding the triacylglycerol propane knuckle; O_2 molar data rather than wt.% data were used to calculate hydrogen demand. Added hydrogen consumption by intentional hydrocracking was calculated at 1.3 wt.% on feed from Pearlson et al.³ and the inputs to each intentional hydrocracking strategy type (Chapter 1), which were taken from the data in Table 8 and used as shown at the end of Table 8 above. Selective-IHC input volume differs among feeds, as described in chapters 1–3.

Hydrogen losses to side-reaction cracking, solubilization in process fluids, and scrubbing and purging of process gases (not shown in Table 8) result in additional hydrogen production, and thus steam reforming emissions. This was addressed for the steam reforming emissions illustrated in Chart 1 by adding 2.5 kg CO_2/b feed to the emissions shown in Table 5, based on steam reforming emissions of 9.82 g CO_2/g H₂ (Chapter 3) and assumed additional hydrogen production of 0.26 kg H₂/b feed. This is a conservative assumption for hydrogen which reflects a lower bound estimate for those losses. Hydrogen losses through side-reaction cracking, solubilization, scrubbing and purging combined would likely range from 102 SCFB (0.26 kg/b) to more than 196 SCFB (0.5 kg/b),² based on analysis of data from a range of published HEFA processing and petroleum processing hydro-conversion process analyses and professional judgment.^{2 4 50–56 93 94 95 96}

References

¹ Data and Methods Table for Feed-specific Estimates (Table 8). Annotated table giving feed-specific data, data sources and analysis methods. Table 8 appears on pages 37–43 above.

² NRDC, 2021. *Changing Hydrocarbons Midstream: Fuel chain carbon lock-in potential of crude-tobiofuel petroleum refinery repurposing;* Natural Resources Defense Council: Washington, D.C. Prepared for the NRDC by Greg Karras, G. Karras Consulting [Needs Link: NRDC link or www.energy-resource.com??? OR "in press"?]

³ Pearlson et al., 2013. A techno-economic review of hydroprocessed renewable esters and fatty acids for jet fuel production. *Biofuels, Bioprod. Bioref.* 7: 89–96. DOI: 10.1002/bbb, 1378.

⁴ Robinson and Dolbear, 2007. Commercial Hydrotreating and Hydrocracking. *In:* Hydroprocessing of heavy oils and residua. Ancheyta, J., and Speight, J., eds. CRC Press, Taylor & Francis Group: Boca Raton, FL. ISBN-13: 978-0-8493-7419-7.

⁵ 2021 Worldwide Refining Survey; Oil & Gas Journal. Capacity data by refinery; includes hydrogen capacities of U.S. refineries as of Dec 2020 (U.S. total refinery hydrogen capacity of 3,578.3 MM cfd converted to MM t/y by the author based on 89.9 kg H2/m3). Accessed Jul 2021 from OGJ website: https://www.ogj.com/ogj-survey-downloads/worldwide-refining/document/14195563/worldwide-us-refinery-surveycapacities-as-of-jan-1-2021

⁶ *Refining Industry Energy Consumption;* table *in* Annual Energy Outlook. U.S. Energy Information Administration: Washington, D.C. CO₂ emissions from total refining industry energy consumption and inputs to distillation units; 2015: 35-AEO2017.4.ref2017-d120816a and 35-AEO2017.25.ref2017-d120816a; 2016: 35-AEO2018.25.ref2018-d121317a and 35-AEO2018.25.ref2018-d121317a; 2017: 35-AEO2019.25.ref2019-d111618a and 35-AEO2019.4.ref2019-d111618a. Data are from the most recent years for which baseline actual data were available as accessed Jul 2021; https://www.eia.gov/outlooks/aeo/data/browser

⁷ Karras, 2010. Combustion Emissions from Refining Lower Quality Oil: What Is the Global Warming Potential? *Environ. Sci. Technol.* 44(24): 9584–9589. *See esp.* Supporting Information, Table S1. https://pubs.acs.org/doi/10.1021/es1019965

⁸ Crude Oil Input Qualities; U.S. Energy Information Administration: Washington, D.C. Accessed Jul 2021 from <u>https://www.eia.gov/dnav/pet/pet_pnp_crq_dcu_nus_a.htm</u>

⁹ Perlack and Stokes, 2011. U.S. Billion-Ton Update: Biomass Supply for Bioenergy and Bioproducts Industry. U.S. Department of Energy, Oak Ridge National Laboratory: Oak Ridge, TN. ORNL/TM-2011/224.

¹⁰ 2020 The State of World Fisheries and Aquaculture. Sustainability in action; Food and Agriculture Organization of the United Nations: Rome. 2020. https://doi.org/10.4060/ca9229en http://www.fao.org/documents/card/en/c/ca9229en

¹¹ Searchinger et al., 2008. Use of U.S. Croplands for Biofuels Increases Greenhouse Gases Through Emissions from Land Use Change. *Science* 319 (5867): 1238-1240. DOI: 10.1126/Science.1151861. https://science.sciencemag.org/content/319/5867/1238

¹² Sanders et al., 2012. *Revisiting the Palm Oil Boom in Southeast Asia;* International Food Policy Research Institute; <u>www.ifpri.org/publication/revisiting-palm-oil-boom-southeast-asia-role-fuel-versus-food-demand-drivers.</u>

¹³ Union of Concerned Scientists USA, 2015. *Soybeans*; <u>www.ucsusa.org/resources/soybeans</u>

¹⁴ Lenfert et al., 2017. *ZEF Policy Brief No. 28;* Center for Development Research, University of Bonn; www.zef.de/fileadmin/user_upload/Policy_brief_28_en.pdf

¹⁵ Nepstad and Shimada, 2018. *Soybeans in the Brazilian Amazon and the Case Study of the Brazilian Soy Moratorium;* Int. Bank for Reconstruction and Development / The World Bank: Washington, D.C. www.profor.info/sites/profor.info/files/Soybeans%20Case%20Study_LEAVES_____

<u>2018.pdf</u>

¹⁶ Santeramo, F., 2017. *Cross-Price Elasticities for Oils and Fats in the US and the EU;* The International Council on Clean Transportation: Beijing, Berlin, Brussels, San Francisco and Washington, D.C. (The ICCT); <u>www.theicct.org/sites/default/files/publications/Cross-price-elasticities-for-oils-fats-US-EU_ICCT_consultant-report_06032017.pdf</u>

¹⁷ Searle, 2017. *How rapeseed and soy biodiesel drive oil palm expansion;* Briefing. The International Council on Clean Transportation: Beijing, Berlin, Brussels, San Francisco and Washington, D.C. (The ICCT); <u>https://theicct.org/publications/how-rapeseed-and-soy-biodiesel-drive-oil-palm-expansion</u>

¹⁸ Takriti et al., 2017. *Mitigating International Aviation Emissions: Risks and opportunities for alternative jet fuels;* The ICCT; <u>https://theicct.org/publications/mitigating-international-aviation-emissions-risks-and-opportunities-alternative-jet</u>

¹⁹ Wang et al., 2016, *Review of Biojet Fuel Conversion Technologies;* NREL/TP-5100-66291. Contract No. DE-AC36-08GO28308. National Renewable Energy Laboratory: Golden, CO. https://www.nrel.gov/docs/fy16osti/66291.pdf

²⁰ See U.S. Environmental Protection Agency, 2016. Lifecycle Greenhouse Gas Results; Overview for Renewable Fuel Standard; and Approved Pathways for Renewable Fuel (palm oil derived hydrotreated diesel does not meet renewable fuel threshold, no approved renewable fuel pathway); https://www.epa.gov/fuels-registration-reporting-and-compliance-help/lifecycle-greenhouse-gas-results https://www.epa.gov/renewable-fuel-standard-program/overview-renewable-fuel-standard https://www.epa.gov/renewable-fuel-standard-program/approved-pathways-renewable-fuel

²¹ Gatti et al., 2021. Amazonia as a carbon source linked to deforestation and climate change. *Nature* 595: 388–393. <u>https://doi.org/10.1038/s41586-021-03629-6</u>

²² Nickle et al., 2021. Renewable diesel boom highlights challenges in clean-energy transition. 3 Mar 2021. *Reuters*. <u>https://www.reuters.com/article/us-global-oil-biofuels-insight-idUSKBN2AV1BS</u>

²³ Walljasper, 2021. GRAINS–Soybeans extend gains for fourth session on veg oil rally; corn mixed. 24 Mar 2021. *Reuters*. <u>https://www.reuters.com/article/global-grains-idUSL1N2LM2O8</u>

²⁴ Kelly, 2021. U.S. renewable fuels market could face feedstock deficit. 8 Apr 2021. *Reuters*. <u>https://www.reuters.com/article/us-usa-energy-feedstocks-graphic/us-renewable-fuels-market-could-face-feedstock-deficit-idUSKBN2BW0EO</u>

²⁵ Portner et al., 2021. IPBES-IPCC co-sponsored workshop report on biodiversity and climate change. IPBES and IPCC. DOI: 10.5281/zenodo.4782538. <u>https://www.ipbes.net/events/launch-ipbes-ipcc-co-sponsored-workshop-report-biodiversity-and-climate-change</u>

²⁶ Diaz et al., 2019. Global Assessment Report on Biodiversity and Ecosystem Services; Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services (IPDES): Bonn, DE. <u>https://ipbes.net/global-assessment</u>

²⁷ Battisti and Naylor, 2009. Historical Warnings of Future Food Insecurity with Unprecedented Seasonal Heat. *Science* 323: 240–244. DOI: 10.1126/science.1164363. https://science.sciencemag.org/content/323/5911/240

²⁸ Wheeler and von Braun, 2013. Climate Change Impacts on Global Food Security. *Science* 341: 508–513. DOI: 10.1126/science.1239402. <u>https://science.sciencemag.org/content/341/6145/508/tab-pdf</u>

²⁹ *How much water is there on, in, and above the Earth?* U.S. Geological Survey: Washington, D.C. <u>https://www.usgs.gov/special-topic/water-science-school/science/how-much-water-there-earth</u>

³⁰ Passow and Carlson, 2012. The biological pump in a high CO₂ world. *Marine Ecology Progress Series* 470: 249–271. DOI: 10.3354/meps09985. <u>https://www.int-res.com/abstracts/meps/v470/p249-271</u>

³¹ Mariani et al., 2020. Let more big fish sink: Fisheries prevent blue carbon sequestration—half in unprofitable areas. *Science Advances* 6(44): eabb4848. <u>https://doi.org/10.1126/sciadv.abb4848</u>

³² Trueman et al., 2014. Trophic interactions of fish communities at midwater depths enhance long-term carbon storage and benthic production on continental slopes. *Proc. R. Soc. B* **281**: 20140669. http://dx.doi.org/10.1098/rspb.2014.0669. <u>https://royalsocietypublishing.org/doi/10.1098/rspb.2014.0669</u>

³³ Low Carbon Fuel Standard Regulation, Title 17, California Code of Regulations, sections 95480– 95503; <u>see</u> esp. § 95488.8(g)(1)(A) and Table 8.

³⁴ *LCFS Pathway Certified Carbon Intensities;* California Air Resources Board: Sacramento, CA. <u>See</u> Current Fuel Pathways spreadsheet under "Fuel Pathway Table," accessed 26 June 2021 at <u>https://ww2.arb.ca.gov/resources/documents/lcfs-pathway-certified-carbon-intensities</u>

³⁵ *Renewable Hydrocarbon Biofuels;* Alternative Fuels Data Center, U.S. Energy Information Administration: Washington, D.C. <u>https://afdc.energy.gov/fuels/emerging_hydrocarbon.html</u>

³⁶ U.S. Supply and Disposition; U.S. Product Supplied of Kerosene-type Jet Fuel; U.S. Energy Information Administration: Washington, D.C. http://www.eia.gov/dnav/pet/pet_sum_snd_d_nus_mbbl_m_cur.htm

³⁷ Share of Liquid Biofuels Produced In-State by Volume; Figure 10 in Low Carbon Fuel Standard Data Dashboard. California Air Resources Board: Sacramento, CA. https://ww3.arb.ca.gov/fuels/lcfs/dashboard/dashboard.htm

³⁸ Schremp (2020). Transportation Fuels Trends, Jet Fuel Overview, Fuel Market Changes & Potential Refinery Closure Impacts. BAAQMD Board of Directors Special Meeting, May 5, 2021, G. Schremp, Energy Assessments Division, California Energy Commission. *In* Board Agenda Presentations Package; <u>https://www.baaqmd.gov/-/media/files/board-of-directors/2021/bods_presentations_050521_revised_op-pdf.pdf?la=en</u>

³⁹ Williams et al., 2012. The Technology Path to Deep Greenhouse Gas Emissions Cuts by 2050: The Pivotal Role of Electricity. *Science* 53–59. <u>https://doi.org/DOI: 10.1126/science.1208365</u>

⁴⁰ Williams et al., 2015. *Pathways to Deep Decarbonization in the United States;* The U.S. report of the Deep Decarbonization Pathways Project of the Sustainable Development Solutions Network and the Institute of Sustainable Development and International Relations. Revision with technical supp. Energy and Environmental Economics, Inc., in collaboration with Lawrence Berkeley National Laboratory and Pacific Northwest National Laboratory. https://usddpp.org/downloads/2014-technical-report.pdf

⁴¹ Williams et al., 2021. Carbon-Neutral Pathways for the United States. *AGU Advances* 2, e2020AV000284. <u>https://doi.org/10.1029/2020AV000284</u>

⁴² Mahone et al., 2020. Achieving Carbon Neutrality in California: PATHWAYS Scenarios Developed for the California Air Resources Board, DRAFT: August 2020; Energy and Environmental Economics, Inc.: San Francisco, CA. <u>https://ww2.arb.ca.gov/sites/default/files/2020-08/e3_cn_draft_report_aug2020.pdf</u>

⁴³ Edwards, 2020. *Jet Fuel Properties;* AFRL-RQ-WP-TR-2020-0017. Fuels & Energy Branch, Turbine Engine Division, Air Force Research Laboratory, Aerospace Systems Directorate, Wright-Patterson Air Force Base, OH, Air Force Materiel Command, U.S. Air Force.

⁴⁴ Oilseeds: World Markets and Trade. Table 42–World vegetable oils supply and distribution, 2013/14– 2020/21; Economic Research Service, U.S. Department of Agriculture, using data from USDA, Foreign Agriculture Service. 26 Mar 2021.

www.ers.usda.gov/webdocs/DataFiles/52218/WorldSupplyUseOilseedandProducts.xlsx?v=5141.3

⁴⁵ *Processed Products–FUS Groups;* data for product type and group 2 name "oil" from NOAA data base. National Oceanographic and Atmospheric Administration. Accessed 13 Jul 2021. https://www.fisheries.noaa.gov/foss/f?p=215:3:10607827382328::NO:::

⁴⁶ Food and Agriculture Organization of the United Nations (FAO) fishery information resource detail, accessed 13 Jul 2021. <u>http://www.fao.org/in-action/globefish/fishery-information/resource-detail/en/c/338773</u>

⁴⁷ *World Data Atlas;* world tallow and lard production in 2018. Accessed 13 Jul 2021. https://knoema.com/data/agriculture-indicators-production+tallow

⁴⁸ U.S. Department of Agriculture *Oil Crops Data: Yearbook Tables; See* tables 5, 20, 26, and 33. <u>https://www.ers.usda.gov/data-products/oil-crops-yearbook/oil-crops-yearbook//#All%20Tables.xlsx?v=7477.4</u>.

⁴⁹ Gerber et al., 2013. *Tackling climate change through livestock—A global assessment of emissions and mitigation opportunities;* Food and Agriculture Organization of the United Nations: Rome. E-ISBN 978-92-5-107921-8 (PDF). <u>http://www.fao.org/news/story/en/item/197623/icode</u>

⁵⁰ Maki-Arvela et al. Catalytic Hydroisomerization of Long-Chain Hydrocarbons for the Production of Fuels. *Catalysts* (**2018**) 8: 534. DOI: 10.3390/catal8110534

⁵¹ Parmar et al. Hydroisomerization of *n*-hexadecane over Brønsted acid site tailored Pt/ZSM-12. *J Porous Mater* (2014). DOI: 10.1007/s10934-014-9834-3

⁵² Douvartzides et al. Green Diesel: Biomass Feedstocks, Production Technologies, Catalystic Research, Fuel Properties and Performance in Compression Ignition Internal Combustion Engines. *Energies* (2019) 12: 809.

⁵³ Regali et al. Hydroconversion of *n*-hexadecane on Pt/silica-alumina catalysts: Effect of metal loading and support acidity on bifunctional and hydrogenolytic activity. *Applied Catalysis* (**2014**) A: General 469: 328. http://dx.doi.org/10.1016/j.apcata.2013.09.048.

⁵⁴ Satyarthi et al. An overview of catalytic conversion of vegetable oils/fats into middle distillates. *Catal. Sci. Technol.* (2013) 3:70. DOI: 10.1039/c2cy20415k. See p. 75.

⁵⁵ Zhao et al., 2017. Review of Heterogeneous Catalysts for Catalytically Upgrading Vegetable Oils into Hydrocarbon Fuels. *Catalysts* 7: 83. DOI: 10.3390/catal7030083. www.mdpi.com/journal/catalysts.

⁵⁶ Tirado et al., 2018. Kinetic and Reactor Modeling of Catalytic Hydrotreatment of Vegetable Oils. *Energy & Fuels* 32: 7245–7261. DOI: 10.1021/acs.energyfuels.8b00947.

⁵⁷ Mahone et al., 2018. *Deep Decarbonization in a High Renewables Future: Updated results from the California PATHWAYS Model;* Report CEC-500-2018-012. Contract No. EPC-14-069. Prepared for California Energy Commission. Final Project Report. Energy and Environmental Economics, Inc.: San Francisco, CA. <u>https://ww2.energy.ca.gov/2018publications/CEC-500-2018-012/CEC-500-2018-012.pdf</u>

⁵⁸ Austin et al., 2021. *Driving California's Transportation Emissions to Zero;* Report No.: UC-ITS-2020-65. Institute of Transportation Studies, University of California. DOI: 10.7922/G2MC8X9X. <u>https://escholarship.org/uc/item/3np3p2t0</u>

⁵⁹ Fact Sheet 2: Sustainable Aviation Fuel: Technical Certification; International Air Transport Association (IATA): Montreal, CA. Accessed Aug 2021 from https://www.iata.org/contentassets/d13875e9ed784f75bac90f000760e998/saf-technical-certifications.pdf

⁶⁰ <u>See</u> "Inherent Safety Measure" requirements to "eliminate hazards to the greatest extent feasible" in California Code of Regulations §§ 5189.1 (c), (*l*) (4) (D), and (*l*) (4) (D).

⁶¹ Tulcan et al., 2008. Analysis of Physical Characteristics of Vegetable Oils. CIGR–International Conference of Agricultural Engineering, Brazil, 31 Aug–4 Sep 2008. https://www.osti.gov/etdeweb/servlets/purl/21512209.

⁶² Han et al., 2013. *Bioresource Technology* 150: 447–456. http://dx.doi.org/10.1016/j.biortech.2013.07.153.

⁶³ Giakoumis, 2018. *Renewable Energy* Vol. 126: 403–419. www.sciencedirect.com/science/article/abs/pii/S0960148118303689.

⁶⁴ Phillips, 2019. Implications of Imported Used Cooking Oil as a Biodiesel Feedstock. NNFCC: Heslington, NY.

⁶⁵ Canale et al., 2005. *Int. J. Materials and Product Technology* 24(1–4): 101–125. https://www.inderscience.com/info/inarticle.php?artid=7943.

⁶⁶ Wang, 2002. *In* Gunstone, ed., Vegetable Oils in Food Technology. Blackwell: Oxford, UK.

⁶⁷ Gunstone, ed., Vegetable Oils in Food Technology. Blackwell: Oxford, UK. 2002.

⁶⁸ After Lindblom, S.C., Dozier, W.A. III, Shurson, G.C., and Kerr, B.J. 2017. Digestibility of energy and lipids and oxidative stress in nursery pigs fed commercially available lipids. J. Anim. Sci. 95: 239-247.

⁶⁹ Shurson et al., 2015. *Journal of Animal Science and Biotechnology* 6:10. DOI: 10.1186/s40104-015-0005-4.

⁷⁰ Kerr et al., 2016. J. Anim. Sci. 94: 2900–2908. doi: 10.2527/jas2016-0440.

⁷¹ Altun et al., 2010. Int. Journal of Engineering Research and Development Vol. 2, No. 2.

⁷² Vingering et al., 2020. *OCL* Vol. 17N° 3 MAI-JUIN 2020. doi: 10.1684/ocl.2010.0309. http://www.ocl-journal.org http://dx.doi.org/10.1051/ocl.2010.0309.

73 Orsavova et al., 2015. Int. J. Mol. Sci. 16: 12871–12890. doi: 10.3390/ijms160612871.

⁷⁴ Awogbemi et al, 2019. *International Journal of Low-Carbon Technologies* 12: 417–425. doi: 10.1093/ijlct/ctz038.

⁷⁵ Rezaei and Azizinejad, 2013. Journal of Food Biosciences and Technology 3.

⁷⁶ Bitman, 1976. *In* Fat Content and Composition of Animal Products: Proceedings of a Symposium. National Academy of Sciences; https://doi.org/10.17226/22.

⁷⁷ Application B0079, Kern Oil & Refining. GREET Pathway for the Production of Renewable Diesel from Animal Tallow. Submitted to Cal. Air Res. Board 31 March 2020.

⁷⁸ Pocket Information Manual, A Buyer's Guide to Rendered Products, National Renderers Association, Inc.: Alexandria, VA. 2003. www.renderers.org. Table e.

⁷⁹ Adapted from Gunstone, F. 1996. Fatty Acid and Lipid Chemistry. Blackie: London, UK.

⁸⁰ Chicken Fat; Fatty Acid Profile. In Material Safety Data Sheet: Chicken Fat. Darling Ingredients Inc.: Irving, TX. Date Prepared: 10 July 2012.

⁸¹ Xie et al., 2019. *Comprehensive Reviews in Food Science and Food Safety* Vol. 18. DOI: 10.1111/1541-4337.12427.

⁸² Gruger, E, 1967. Fatty Acid Composition of Fish Oils. U.S. Dept. of Interior, Fish and Wildlife Service, Bureau of Commercial Fisheries: Washington, D.C. https://spo.nmfs.noaa.gov/content/circular-276-fatty-acid-composition-fish-oils.

⁸³ Moffat and McGill, Ministry of Agriculture, Fisheries and Food: Torry Research Station, Aberdeen AB9 8DG. 1993. Variability of the composition of fish oils: significance for the diet. *Proceedings of the Nutrition Society* 52: 441–456. Printed in Great Britain. *After* Ackman and Eaton, 1966; Jangaard et al., 1967.

⁸⁴ Suseno et al., 2014. Fatty Acid Composition of Some Potential Fish Oil from Production Centers in Indonesia. *Oriental Journal of Chemistry* 30(3): 975–980. http://dx.doi.org/10.13005/ojc/300308.

⁸⁵ Simat et al., 2019. Production and Refinement of Omega-3 Rich Oils from Processing By-Products of Farmed Fish Species. *Foods* 8(125). doi: 10.3390/foods8040125.

⁸⁶ EUBIA, *after* Wen et al., 2010. http://www.eubia.org/cms/wiki-biomass/biomass-resources/challenges-related-to-biomass/used-cooking-oil-recycling.

⁸⁷ Knothe and Steidly, 2009. *Bioresource Technology* 100: 5796–5801. doi: 10.1016/j.biortech.2008.11.064.

⁸⁸ Banani et al., 2015. *J. Mater. Environ. Sci.* 6(4): 1178–1185. ISSN: 2028–2508. CODEN: JMESCN. http://www.jmaterenvironsci.com.

⁸⁹ Chhetri et al., 2008. *Energies* 1: 3–8. ISSN 1996-1073. www.mdpi.org/energies. DOI: 10.3390/en1010003.

⁹⁰ Yusuff et al., 2018. Waste Frying Oil as a Feedstock for Biodiesel Production. IntechOpen http://dx.doi.org/10.5772/intechopen.79433.

⁹¹ Mannu et al., 2019. Variation of the Chemical Composition of Waste Cooking Oils upon Bentonite Filtration. *Resources* 8 (108). DOI: 10.3390/resources8020108.

⁹² Mishra and Sharma, 2014. *J Food Sci Technol* 51(6): 1076–1084. DOI: 10.1007/s13197-011-0602-y.

⁹³ Speight, J. G., 1991. The Chemistry and Technology of Petroleum; 2nd Edition, Revised and Expanded. *In* Chemical Industries, Vol. 44. ISBN 0-827-8481-2. Marcel Dekker: New York. <u>See</u> pp. 491, 578–584.

⁹⁴ Speight, J. G., 2013. Heavy and Extra-heavy Oil Upgrading Technologies. Elsevier: NY. ISBN: 978-0-12-404570-5. pp. 78–79, 89–90, 92–93.

⁹⁵ Meyers, R. A., 1986) Handbook of Petroleum Refining Processes. *In* Chemical Process Technology Handbook Series. ISBN 0-07-041763-6. McGraw-Hill: NY. *See* pp. 5-16 and 5-17.

⁹⁶ Bouchy et al., 2009. Fischer-Tropsch Waxes Upgrading via Hydrocracking and Selective Hydroisomerization. *Oil & Gas Science and Technology—Rev.* 64(1): 91-112. DOI: 10.2516/ogst/2008047.